InfoServer 1000

Installation and Owner's Guide

Order Number: EK-INFLC-OM. B01

This manual describes how to install, use, test, and maintain the InfoServer 1000 system and packaged configurations.

Revision/Update Information:	This is a revision. This guide supersedes InfoServer 1000 Installation and Owner's Guide; Order No: EK-INLFC-OM. A01
Software Version:	InfoServer Version 3.1

Digital Equipment Corporation Maynard, Massachusetts

First Edition, April 1993 Revision, March 1994

Digital Equipment Corporation makes no representations that the use of its products in the manner described in this publication will not infringe on existing or future patent rights, nor do the descriptions contained in this publication imply the granting of licenses to make, use, or sell equipment or software in accordance with the description.

Possession, use, or copying of the software described in this publication is authorized only pursuant to a valid written license from Digital or an authorized sublicensor.

© Digital Equipment Corporation 1993, 1994. All rights reserved.

FCC NOTICE: The equipment described in this manual generates, uses, and may emit radio frequency energy. The equipment has been type tested and found to comply with the limits for a Class A computing device pursuant to Subpart J of Part 15 of FCC Rules, which are designed to provide reasonable protection against such radio frequency interference when operated in a commercial environment. Operation of this equipment in a residential area may cause interference, in which case the user at his own expense may be required to take measures to correct the interference.

The prepaid READER'S COMMENTS form at the end of this document requests the user's critical evaluation to assist in preparing future documentation.

The following are trademarks of Digital Equipment Corporation: DECconnect, DECstation, DELNI, DEMPR, InfoServer, OpenVMS, ThinWire, ULTRIX, VAX, VAX DOCUMENT, VAXstation, VT300, and the DIGITAL logo.

Digital Equipment Corporation has applied for the following trademarks: InfoServer Librarian, InfoServer Publisher, InfoServer Scribe, InfoServer System Manager, InfoServer VXT, and InfoTower.

The following are third-party trademarks:

Macintosh is a registered trademark of Apple Computer, Inc.

MS-DOS is a registered trademark of Microsoft Corporation.

All other trademarks and registered trademarks are the property of their respective holders.

This document was prepared using VAX DOCUMENT Version 2.1.

Für Bundesrepublik Deutschland For Federal Republic of Germany Pour la République féderale d'Allemagne

Hochfrequenzgerätezulassung und Betriebsgenehmigung

Bescheinigung des Herstellers/Importeurs:

Hiermit wird bescheinigt, daß die Einrichtung in Übereinstimmung mit den Bestimmungen der DBP-Verfügung 523/1969, Amtsblatt 113/1969, und Grenzwertklasse "A" der VDE0871, funkentstört ist.

Das Bundesamt für Zulassungen in der Telekommunikation (BZT), hat diesem Gerät eine EMV-Zulassungsnummer zugeteilt.

Betriebsgenehmigung:

Hochfrequenzgeräte dürfen erst in Betrieb genommen werden, nachdem hierfür von der für den vorgesehenen Aufstellungsort zuständigen Außenstelle des Bundesamtes für Post und Telekommunikation (BAPT) die Genehmigung erteilt ist.

Als Antrag auf Erteilung einer Genehmigung dient eine Anmeldepostkarte (Anhang des Handbuches) mit Angabe der EMV-Zulassungsnummer.

Der untere Teil der Anmeldepostkarte ist vom Betreiber zu vervollständigen und an die örtliche Außenstelle des Bundesamtes für Post und Telekommunikation (BAPT) zu schicken. Der obere Teil bleibt beim Gerät.

Betreiberhinweis:

Das Gerät wurde funktechnisch sorgfältig entstört und geprüft. Die Kennzeichnung mit der Zulassungsnummer bietet Ihnen die Gewähr, daß dieses Gerät keine anderen Fernmeldeanlagen einschließlich Funkanlagen stört.

Sollten bei diesen Geräten ausnahmsweise trotzdem, z.B. im ungünstigsten Fall beim Zusammenschalten mit anderen EDV-Geräten, Funkstörungen auftreten kann das im Einzelnen zusätzliche Funkentstörungsmaßnahmen durch den Benutzer erfordern.

Bei Fragen hierzu wenden Sie sich bitte an die örtlich zuständige Außenstelle Ihres Bundesamtes für Post und Telekommunikation (BAPT).

Externe Datenkabel:

Sollte ein Austausch der von Digital spezifizierten Datenkabel nötig werden, muß der Betreiber für eine einwandfreie Funkentstörung sicherstellen, daß Austauschkabel im Aufbau und Abschirmqualität dem Digital Originalkabel entsprechen.

Kennzeichnung:

Die Geräte werden bereits in der Fertigung mit der Zulassungsnummer gekennzeichnet und mit einer Anmeldepostkarte versehen. Sollte Kennzeichnung und Anmeldepostkarte übergangsweise nicht mit ausgeliefert werden kontaktieren Sie bitte das nächstgelegene Digital Equipment Kundendienstbüro. この装置は、第一種情報装置(商工業地域において使用されるべき情報装置) で商工 業地域での電波障害防止を目的とした情報処理装置等電波障害自主規制協議会 (VCCI) 基準に適合しております。

従って,住宅地域またはその隣接した地域で使用すると, ラジオ,テレビジョン受信 機等に受信障害を与えることがあります。

取扱説明書に従って正しい取り扱いをして下さい。

Contents

P	reface .		xiii
1	InfoSer	ver 1000 System Overview	
	1.1	Introduction	1-1
	1.2	System Highlights	18
	1.3	CD-ROM Services	1–9
	1.4	Console Terminal	1–10
	1.5	Front Panel Indicators	1–10
2	Installi	ng Standalone System Configurations	
	2.1	Introduction	2–1
	2.2	Choosing the Right System Location	2–1
	2.3	Unpacking and Inspecting the Standalone System	
		Components	2–3
	2.3.1	Connecting the System Components	2–5
	2.3.2	Identifying the InfoServer 1000 Ports and Connectors	2–5
	2.3.3	Attaching the Ethernet Terminator or Loopback Plug	2–5
	2.3.3.1	Connecting the ThinWire Ethernet Terminator	2–7
	2.3.3.2	Connecting the Standard Ethernet Loopback Plug	2–9
	2.3.4	Connecting a Console Terminal	2–10
	2.3.5	Connecting the Standalone Expansion Peripherals	2-12
	2.3.6	Connecting the Standalone InfoServer 1000 Power Supply	
		and Power Cord	2–12
	2.4	Turning On and Off the System Power	2–14

3 Installing InfoTower System Configurations

3.1	Introduction	3–1
3.2	Choosing the Right System Location	3–1
3.3	Unpacking and Inspecting InfoTower System Components	3–2
3.3.1	Connecting the System Components	3–6
3.3.2	Identifying the InfoTower Ports and Connectors	3–6
3.3.3	Installing the SCSI Terminator	3–8
3.3.4	Attaching the Ethernet Terminator or Loopback Plug	3–9
3.3.4.1	Connecting the ThinWire Ethernet Terminator	3–10
3.3.4.2	Connecting the Standard Ethernet Loopback Plug	3–12
3.3.5	Connecting a Console Terminal	3–13
3.3.6	Connecting System Power to InfoTower Configurations	315
3.4	Turning On and Off the System Power	3–17

4 Installing StorageWorks System Configurations

4.1	Introduction	4-1
4.2	Choosing the Right System Location	4–1
4.3	Unpacking and Inspecting the StorageWorks System	
	Components	4–2
4.4	Installing the InfoServer SW1000	4-4
4.4.1	Identifying InfoServer SW1000 Ports and Connectors	4-4
4.4.2	Determine SCSI Bus Configuration and Termination	4–5
4.4.3	Insert InfoServer SW1000 into StorageWorks Pedestal	4-9
4.4.3.1	Installing a Second 5.25-Inch Half-Height Device	4–10
4.4.4	Installing a Pre-Configured InfoServer SW1000	4–11
4.4.5	Connecting a Console Terminal	4–12
4.4.6	Attaching the Ethernet Terminator	4–13
4.4.6.1	Connecting the ThinWire Ethernet Terminator	4–14
4.5	Turning On and Off the System Power	4–15

5 Completing the System Installation

5.1	Introduction	51
5.2	Starting the System	5–1
5.3	Connecting the System to a Network	5–2
5.4	Booting the System	5–3
5.4.1	Booting the System Automatically Without a Console	53
5.4.2	Booting the System with a Console (If Necessary)	5–3

6 Installing Expansion Peripherals

6.1	Introduction	6-1
6.2	Guidelines for Connecting Expansion Peripherals	6–2
6.3	Unpacking an Expansion Peripheral	6–2
6.4	Connecting One Expansion Peripheral	6–3
6.5	Connecting Two Expansion Peripherals	6–5
6.6	CD-ROM Compact Disc Expansion Peripheral	6–7
6.6.1	Verifying the SCSI ID Number on the CD-ROM Expansion	
	Peripheral	67
6.6.2	Determining Available SCSI ID's	6–8
6.7	Preventive Maintenance	69

7 Connecting to a Network

7.1	Introduction	7–1
7.2	Connecting to a ThinWire Ethernet Network	7–2
7.2.1	Verifying Your ThinWire Ethernet Network Adapter	72
7.2.2	Connecting Your System to a ThinWire Ethernet Cable	7–3
7.3	Creating a ThinWire Ethernet Workgroup	7–5
7.3.1	Connecting Your System to an OPEN DECconnect	
	Faceplate	7–5
7.3.2	Troubleshooting the ThinWire Segment	7–7
7.4	Connecting to a Standard Ethernet Network	7–9
7.4.1	Verifying Your Standard Ethernet Network Adapter	7–9
7.4.2	Connecting a Transceiver Cable to Your System	7–10
7.4.3	Troubleshooting the Standard Ethernet Connection	7–11

8 Solving Problems

8.1	Introduction	8–1
8.2	Checking the Cable Connections	8–1
8.3	Using the Troubleshooting Table	8–3
8.4	Entering the Console Mode	8–5
8.5	Interpreting the Power-On Error Messages	8–5
8.6	Running the Self-Tests	8–7
8.7	Displaying the System Configuration Information	89
8.8	Displaying the Device Information	8–10
8. 9	Running the Test Utilities for Expansion Peripherals	8-11
8.10	Restarting the System After Running Tests	8-11
8.11	Reestablishing the Console Parameters	8–12
8.12	Recovering from Bootstrap Failures	8-12
8.13	Service Information	8-17

	8.14	Command Summary	8–17
A	Install	ation Checklist	
	A.1 A.2 A.3 A.4	Introduction	A1 A1 A2 A3
В	Loadir	ng and Unloading Compact Disks	
	B.1 B.2 B.3	Introduction	B–1 B–1 B–6
С	C SCSI ID Numbers		
	C.1 C.2 C.3	Introduction	C1 C1 C3
D	Syster	m Specifications	
	D.1 D.2 D.3 D.4 D.5	Introduction InfoServer 1000 Hardware Specifications Standalone InfoServer 1000 Specifications InfoTower System Specifications InfoServer SW1000 Specifications	D-1 D-2 D-6 D-9
Ε	Field-I	Replaceable Units	
	E.1	Introduction	E-1
GI	ossary		

Index

Figures

1–1	Standalone InfoServer 1000	1–2
1–2	The InfoServer Librarian	1–3
1–3	The InfoServer VXT	14
14	The InfoServer Scribe	1–5
1–5	The InfoServer SW1000 SBB	1–6
1–6	The InfoServer SW1000 in a StorageWorks Pedestal	1–7
17	InfoServer 1000 Front Panel Indicators	1–10
2–1	Standalone InfoServer 1000 System Components	2–4
2–2	Standalone InfoServer 1000 Ports and Connectors	2-6
23	Attaching the Terminators to the T-Connector	2–7
2–4	Connecting the T-Connector	28
2–5	Connecting the Standard Ethernet Loopback Plug	2–9
2–6	Connecting the Console Terminal Cable	2–11
2–7	Connecting InfoServer 1000 AC and DC Power Cords	2–13
3–1	System Components	3–4
3–2	Properly Seating Peripherals	35
3–3	InfoTower Ports and Connectors	3–7
3–4	Installing a SCSI Terminator on an InfoTower Rear Panel	
	•••••••••••••••••••••••••••••••••••••••	38
3–5	Attaching the Terminators to the T-Connector	3–10
3–6	Connecting the T-Connector	3–11
3–7	Connecting the Standard Ethernet Loopback Plug	3–12
38	Connecting the Console Terminal Cable	3–14
3–9	Connecting the InfoTower Enclosure Power Cord	3–16
4–1	InfoServer SW1000 System Components	4–3
4–2	InfoServer SW1000 Ports and Connectors	4-5
43	Single Shelf, Single Bus Configuration	46
4-4	Single Shelf, Two Buses Configuration	48
4–5	Double Shelf, Single Bus Configuration	4–9
4–6	Inserting the SBB into StorageWorks Pedestal	4–10
4–7	StorageWorks Pedestal with InfoServer SW1000	4–11
4–8	Connecting the Console Terminal Cable	4–13
4-9	Attaching the Terminators to the T-Connector	4–14

4–10	Connecting the T-Connector	4–15
6–1	Connecting One CD–ROM Expansion Peripheral	6-4
62	Connecting Two CD-ROM Expansion Peripherals	66
6–3	CD-ROM Rear Panel	6–7
7–1	Attaching the T-Connector with Cable Segment to the BNC	
	Connector	7–4
7–2	Connecting to the OPEN DECconnect Faceplate	7–6
7–3	Disconnecting the T-Connector	7–8
74	Securing the Standard Ethernet Cable	711
8-1	Setting the External Jumper (J1)	8–14
B–1	Opening the Caddy	B–3
B–2	Loading the Disc	B-4
B–3	Loading the Caddy into the Drive	B5
B4	Ejecting the Caddy from the Drive	B6
B–5	Front Panel of the Drive	B–7

Tables

Related Documents	xv
Optional Storage Drives	19
SCSI Bus Parameters	4–7
Troubleshooting Basic System Problems	8–3
Power-On Display and Self-Test Identifiers	88
SET Command Summary	8–17
SHOW Command Summary	8–18
SCSI ID Numbers for InfoServer 1000 System	C-2
InfoServer 1000 System Hardware Specifications	D-1
Standalone InfoServer 1000 Power Specifications	D-2
Standalone InfoServer 1000 Weight and Dimensions	D-2
Standalone InfoServer 1000 Operating Conditions	D-3
Standalone InfoServer 1000 Storage Conditions	D3
InfoServer 1000 System Acoustic Values (Declared Values Per ISO 9296 and ISO 7779)	D4
InfoServer 1000 Schallemissionswerte (Werteangaben nach	
ISO 9296 und ISO 7779/DIN EN27779)	D5
InfoTower System Functional Specifications	D6
InfoTower System Physical Specifications	D-7
	Related DocumentsOptional Storage DrivesSCSI Bus ParametersTroubleshooting Basic System ProblemsPower-On Display and Self-Test IdentifiersSET Command SummarySHOW Command SummarySCSI ID Numbers for InfoServer 1000 SystemInfoServer 1000 System Hardware SpecificationsStandalone InfoServer 1000 Power SpecificationsStandalone InfoServer 1000 Operating ConditionsStandalone InfoServer 1000 Operating ConditionsStandalone InfoServer 1000 System Acoustic Values (Declared ValuesPer ISO 9296 and ISO 7779)InfoServer 1000 Schallemissionswerte (Werteangaben nachISO 9296 und ISO 7779/DIN EN27779)InfoTower System Functional SpecificationsInfoTower System Physical Specifications

D–10	InfoTower System Input Characteristics—Electrical Specifications	D7
D–11	InfoTower System Output Characteristics—Electrical Specifications	D8
D-12	InfoTower System Acoustic Values (Declared Values Per ISO 9296 and ISO 7779)	D8
D–13	InfoTower Schallemissionswerte (Werteangaben nach ISO 9296 und ISO 7779/DIN EN27779)	D-9
D-14	InfoServer SW1000 Weight and Dimensions	D-9
D15	InfoServer SW1000 Power Specifications	D-10
D-16	InfoServer SW1000 System Acoustic Values (Declared Values Per ISO 9296 and ISO 7779)	D-10
D17	InfoServer SW1000 Schallemissionswerte (Werteangaben nach ISO 9296 und ISO 7779/DIN EN27779)	D-10
D–18	Storage Works InfoServer SW1000 Operating Conditions	D-11
D-19	StorageWorks InfoServer SW1000 Storage and Nonoperating Conditions	D–11
D-20	StorageWorks InfoServer SW1000 Thermal Stabilization Specifications	D-12
E-1	InfoServer 1000 Field-Replaceable Units	E–2
E2	InfoTower Field-Replaceable Units	E3
E3	AC Power Cables	E–3

.

• , .

Preface

Document Purpose

This document describes how to install, use, test, and maintain the InfoServer 1000 system, the related InfoServer SW1000 system, and packaged configurations including either as a component. The InfoServer is a generalpurpose disk storage server that quickly and efficiently transfers data between SCSI devices connected to the server and remote network client systems.

For information on session management and software commands, see the *InfoServer System Operations Guide*.

Intended Audience

This document is designed for the people responsible for installing, using, or maintaining InfoServer 1000-based systems.

Document Structure

This document is organized as follows:

- Chapter 1 provides an overview of the InfoServer 1000-based systems, including the available configurations and indicator descriptions.
- Chapter 2 provides instructions for installing standalone InfoServer 1000 systems.
- Chapter 3 provides instructions for installing InfoTower mounted InfoServer 1000 systems.
- **Chapter 4** provides instructions for installing new InfoServer SW1000 systems.
- **Chapter 5** provides instructions for completing an installation, including the starting and booting of InfoServer 1000-based systems.
- Chapter 6 provides information on installing and using add-on storage devices and software-loading devices such as compact-disc drives.

- **Chapter 7** explains how to connect your system to a ThinWire Ethernet or a standard Ethernet network. The chapter also explains how to connect systems into a simple ThinWire workgroup.
- Chapter 8 provides troubleshooting information.

Appendixes provide information on the following topics:

- Appendix A provides a checklist of the basic installation steps.
- Appendix B provides instructions for loading and unloading CD-ROM drives.
- Appendix C provides instructions for setting SCSI ID numbers on drives for custom configurations.
- Appendix D provides system specifications.
- Appendix E lists system field replaceable units.

The Glossary defines technical terms used in the document.

Conventions

The following conventions are used in this document:

Return	Boxed wording represents a key name. It indicates that you press a key on the keyboard.
10	This specifically represents the on/off power switch on equipment. You press to turn power on, or O to turn power off.
boldface text	Boldface text represents the introduction of a new term defined in the glossary, or a phrase that is emphasized in text.
UPPERCASE TEXT	Uppercase letters indicate that you must enter a command (for example, enter TEST 50).
numbers	Unless otherwise noted, all numbers in the text are assumed to be decimal. Nondecimal radixes—binary, octal, or hexadecimal—are explicitly indicated.

Related Documents

Consult your Digital sales representative for information on obtaining the documents listed in Table 1.

Table 1 Related Documents

Title	Order Number			
InfoServer Documentation				
InfoServer System Operations Guide	AA-PJXJB-TE			
InfoServer 100 Installation and Owner's Guide	EK-DIS1K-IN			
InfoServer 150 Installation and Owner's Guide	EK-INFSV-OM			
InfoServer 1000 Installation and Owner's Guide	EK-INFLC-OM			
InfoTower Installation and Owner's Guide	EK-INFOE-OM			
Client Documentation				
InfoServer Client for OpenVMS LASTCP and LADCP Utilities	AA-PT4DA-TE			

InfoServer Client for OpenVMS LASTCP and LADCP Utilities	s AA-PT4DA-TE
LASTport Handbook	EK-LADLA-AS not available
InfoServer Client for Macintosh Installation and User's Guide	AA-PPV5B-TE
InfoServer Client for DOS Installation and User's Guide	AA-PJX
InfoServer Client for ULTRIX Installation and User's Guide	AA-PNRDA-TE
InfoServer Software Version 3.1 SPD	33.20.07
	(continued on next page)

99 Joinder 04-01 QA-YSHAD-DOS X HW \$250-

AA-PJX98-TE Overvide on code?

Title	Order Number
SCSI Device Documentation	
RRD40 Disc Drive Owner's Manual	EK-RRD40-OM
RRD42 Disc Drive Owner's Manual	EK-RRD42-OM
RRD43 Disc Drive Owner's Manual	EK-RRD43-IN
RRD44 CD-ROM Installation Guide	EK-RRD44-IN
RWZ01 Magneto-Optical Disk Drive User's Guide	EK-RWZ01-UG
RZ24 Hard Disk Drive Installation Guide	EK-RZ24I-IS
SZ03 Storage Expansion Box Owner's Guide	EK-SDDBY-OG
TK50Z User's Guide	EK-OTK50-UG
TLZ04 Cassette Tape Drive Owner's Manual	EK-TLZ04-OM
TLZ06 Cassette Tape Drive Installation Guide	EK-STEXP-AD
TLZ06 Cassette Tape Drive Owner's Manual	EK-TLZ06-OM
TZ85 Tabletop Cartridge Tape Subsystem Installation Guide	EK-TZ85T-IG
TZK10 Cartridge Tape Drive Owner's Guide	EK-TZK10-OG
StorageWorks Documentation	
StorageWorks Family 5 1/4-Inch SBB Half-Height Device Installation Guide	EK-SBB55-IG
StorageWorks Family User's Guide	EK-BA350-UG
StorageWorks Family Configuration Guide	EK-BA350-CG
DECconnect System Documentation	
DECconnect System General Description	EK-DECSY-GD
DECconnect System Requirements Evaluation Workbook	EK-DECSY-EG
DECconnect System Installation and Verification Guide	EK-DECSY-VG
DEC connect System Standalone ThinWire Networks: Planning and Installation Guide	EK-DECSY-TG
DECconnect System Planning and Configuration Guide	EK-DECSY-CG

Table 1 (Cont.) Related Documents

1

InfoServer 1000 System Overview

1.1 Introduction

The InfoServer 1000 system is a dedicated SCSI storage server that quickly and efficiently transfers data between SCSI devices connected to the server and remote network **client systems**. It provides a low-cost data sharing capability that is flexible and easy to manage.

The InfoServer 1000 system can support many diverse operating systems and applications. For example, OpenVMS, ULTRIX, MS-DOS, Macintosh, and other client systems can share software distribution and documentation on compact discs connected to the server. Most VAX systems can use the server to load their **bootstrap** program, and then perform an OpenVMS operating system installation from a compact disc connected to the server.

The InfoServer 1000 is a small, modular black box server. When the server is attached through SCSI cabling to specific storage devices, it offers the devices to clients on the network. Thus, the clients can share access requests to the devices. Designed with an industry-standard 5.25-inch form factor, it can either sit on a table top daisy-chained to devices, be mounted into an InfoTower enclosure, or be integrated into other 5.25-inch storage bays like StorageWorks System Building Blocks (SBBs).

InfoServer 1000 System Overview

The InfoServer 1000 may be configured with many different storage options. Figure 1–1 shows the standalone InfoServer 1000 product with its external power supply. This is the main component of the server system.

Figure 1–1 Standalone InfoServer 1000



InfoServer Librarian

Figure 1-2 shows the InfoServer 1000 mounted in an InfoTower enclosure. The InfoTower enclosure houses the InfoServer 1000 along with up to seven additional SCSI storage devices. This figure depicts the InfoServer Librarian configuration. The InfoServer Librarian is designed for users needing to share CD-ROM applications in a network. It is configured with either four or seven CD-ROM drives.

Figure 1–2 The InfoServer Librarian



MK331-02

InfoServer VXT

Figure 1-3 shows the InfoServer VXT configuration. The InfoServer VXT is designed for X-Window terminal users desiring to reduce expense and off-load the host through the use of a server. It is configured with a fixed hard-disk drive and an optional CD-ROM drive.

Figure 1–3 The InfoServer VXT



InfoServer Scribe

Figure 1-4 shows the InfoServer Scribe configuration. The InfoServer Scribe is designed for users desiring to back up their systems over the network onto tape subsystems. It is configured with either a high-capacity cartridge tape drive (shown) or a 4 mm digital audiotape (DAT) drive.

Figure 1-4 The InfoServer Scribe



InfoServer SW1000

Figure 1-5 shows the InfoServer SW1000 StorageWorks building block (SBB). The InfoServer SW1000 is designed to be used with the StorageWorks family of products. Its 5.25-inch system building block form-factor enables it to be integrated into StorageWorks shelves, where other StorageWorks devices also reside.

Figure 1–5 The InfoServer SW1000 SBB



InfoServer SW1000 is also integrated into a factory-configured StorageWorks pedestal or shelf as shown in Figure 1-6.





1.2 System Highlights

Your InfoServer 1000 system has many special features.

- Memory—The system comes with 4 MB of internal random access memory (RAM).
- ThinWire Ethernet or standard Ethernet connectability—You can connect the InfoServer 1000 system to ThinWire or standard Ethernet workgroups and networks, but not to more than one type at the same time. InfoServer SW1000 only comes with the ThinWire Ethernet connection.
- Internal writeable configuration storage—The InfoServer 1000 system includes internal memory that stores server configuration information.
- Self-sensing power supply—The external power supply (standalone configuration) accepts power inputs in the ranges of 100 Vac to 120 Vac and 200 Vac to 240 Vac, at 50 or 60 Hz. A selection switch is not needed.

The InfoTower power supply configuration is switch selectable.

- Communication ports—A single DEC423 asynchronous port (with MMJ connector for VT-series terminals) is a standard provision. You can use this port to connect a terminal to the server to establish a local server management session. (Refer to Section 2.3.4 and the *InfoServer System Operations Guide* for more information.) This port is preset to run at 9600 baud. Any terminal connected to this port must initially be set at 9600 baud.
- SCSI connectability—The Small Computer Systems Interconnect II (SCSI-II) is an industry standard for connecting mass storage devices.
- Add-on storage devices—External standalone expansion peripherals may be purchased for the standalone configuration to provide added storage capability.

These external peripherals must be qualified by Digital to meet the system requirements. Table 1–1 is a partial list of supported devices. Refer to the InfoServer Software, Software Product Description (SPD 33.20.07) for a complete list of supported devices. You can connect up to seven (7) external drives to the InfoServer 1000, provided that you observe SCSI bus length guidelines.

You can also purchase an InfoTower enclosure that houses and connects the InfoServer 1000 along with up to seven (7) *internal* drives. The InfoTower provides the benefits of less power consumption (one power supply for all drives), reduced desktop space, and elimination of cable clutter.

The InfoServer 1000 can also be integrated into the StorageWorks pedestals. InfoServer SW1000 enables the StorageWorks devices to be shared by clients in multivendor networks without other servers.

Model	Description	Capacity
RRD40	Compact disc drive	600 MB
RRD42	Compact disc drive	600 MB
RRD43	Compact disc drive	600 MB
RRD44	Compact disc drive	600 MB
RWZ01	Magneto-optical disk drive	594 MB
RZ23L	Hard-disk drive	121 MB
RZ24L	Hard-disk drive	245 MB
RZ25	Hard-disk drive	426 MB
RZ26	Hard-disk drive	1.05 GB
RZ56	Hard-disk drive	665 MB
RZ57	Hard-disk drive	1 GB
RZ58	Hard-disk drive	1.38 GB
TK50Z	Cartridge tape drive	Up to 95.0 MB
TLZ04	4 mm digital audiotape drive	Up to 1.2 GB
TLZ06	4 mm digital audiotape drive	Up to 4.0 GB
TZ85	Cartridge tape drive	Up to 2.6 GB
TZK10	Cartridge tape drive	Up to 525 MB
TZK11	Cartridge tape drive	Up to 2 GB

Table 1–1 Optional Storage Drives

1.3 CD-ROM Services

Any InfoServer 1000 system should have at least one compact-disc drive configured in the system or network for software updating purposes.

CD-ROM services will not be evident on the network unless there is a compact disc loaded on one of the CD-ROM readers attached to the InfoServer 1000 system.

.

1.4 Console Terminal

Design of the InfoServer 1000 system requires a minimum amount of system management. You normally issue server commands by using a LAT terminal session created to the InfoServer 1000 system from a terminal server. In the rare event that the InfoServer 1000 has problems communicating with client systems, you can connect a terminal to the MMJ port on the back of the enclosure (refer to Section 2.3.4) and follow the instructions in Chapter 8 to troubleshoot the problem. The recommended console terminal is a VT300 or VT400 series terminal.

1.5 Front Panel Indicators

The InfoServer 1000 has six front panel indicators. They show the status of various system conditions. Figure 1–7 shows the indicator locations and names. A list of each indicator and its function follows. Familiarize yourself with the meaning of these indicators before proceeding to system installation and startup.

Figure 1–7 InfoServer 1000 Front Panel Indicators



INDICATOR FUNCTIONS

System Group



Power OK—The H7827-BA, InfoTower power supply, or the StorageWorks BA35X-HA is providing the correct DC voltages to the InfoServer 1000. The InfoServer 1000 is turned on.

System OK—When blinking once/second indicates that the power-on self-test diagnostics have passed and the InfoServer software kernel is running. If on or off, the software is halted or hung.

Network Group

Network OK—When on or blinking, the InfoServer 1000 is connected to an operating Ethernet and there is network activity.

---> Network Activity—When on, the unicast network packet received is destined for this particular InfoServer 1000. This does not apply to multicast packets.

SCSI Group

- SCSI OK—When on, all SCSI peripherals on the bus are accessible. When off, one or more SCSI peripherals on the bus are not accessible (perhaps powered off).
- ---> SCSI Activity-When on, a data transfer is in process on the SCSI bus. When off, no data of any kind is being passed on the SCSI bus.

.

2

Installing Standalone System Configurations

2.1 Introduction

This chapter covers the hardware installation procedures for InfoServer 1000 packaged system configurations, such as, an InfoServer Local Area CD.

Installation procedures cover the standalone InfoServer 1000 and use the compact-disk drive as an example of an expansion peripheral. For specific installation procedures and operating instructions for other peripherals, refer to the documentation provided with the peripheral.

Installing and setting up your standalone InfoServer 1000 system is easy. This chapter provides step-by-step instructions on how to perform the following operations:

- Choosing the right location for the system
- Unpacking, inspecting, and connecting system components
- Turning on and off the system power

Appendix A contains a brief installation checklist, without the details provided in this chapter.

2.2 Choosing the Right System Location

Use the following checklist to keep your system operating at its best:

- Carefully read all installation instructions before you turn on the power.
- Keep the temperature between 10° C and 40° C (50° F and 90° F).
- Keep the relative humidity between 10 percent and 80 percent.

- Keep the air well circulated to prevent excess heat and dust from accumulating.
- Keep your equipment away from heaters, photocopiers, direct sunlight, and abrasive particles.
- Before you set up your system, select a surface that will hold the standalone InfoServer 1000 and associated expansion peripherals. Your desk or work table is a good choice.
- Keep the area clean. Do not place food or liquid on or near your equipment, and do not place your InfoServer 1000 directly on the floor. Dust and dirt will damage the system components.
- Connect your InfoServer 1000 system to a dedicated grounded circuit.

If you have several pieces of equipment that need to be plugged into a power outlet, use a grounded power strip. Many power strips come with an on/off switch and a power-surge protector.

Do not exceed the circuit power capacity.

• To avoid damaging equipment that has been moved inside from a cold environment, let the equipment warm to room temperature before you turn it on.

2.3 Unpacking and Inspecting the Standalone System Components

The standalone InfoServer 1000 system is shipped in one carton. This carton contains the InfoServer 1000 hardware with accessories and documentation. The smaller carton, packed inside the larger carton, contains the software compact discs with licenses and instructions. Standalone expansion peripherals are shipped in separate cartons.

____ CAUTION _____

InfoServer 1000 software is factory-loaded. The software compact discs shipped with the system are for backup purposes. Do not attempt to load software as part of the installation process.

Figure 2-1 shows the components shipped with the standalone InfoServer 1000 system. Check the contents of the cartons against the shipping list provided. Expansion storage devices are shipped in separate cartons (refer to Section 6.3).

Note _

Depending on how you set up your system, you may not use all the components supplied with your system. Store any unused parts safely for future use.

Moving the System

Always repack the InfoServer 1000 in its original packing material when moving or relocating your system. This precaution will protect the hardware from damage.

Figure 2–1 Standalone InfoServer 1000 System Components



MKV-040000331-42

2.3.1 Connecting the System Components

After unpacking, follow the instructions in Sections 2.3.2, 2.3.3, 2.3.4, 2.3.5, and 2.3.6 to connect system components.

2.3.2 Identifying the InfoServer 1000 Ports and Connectors

Figure 2-2 shows all the ports and connectors on the rear of the InfoServer 1000 system. Take some time to become familiar with the ports and connectors. Symbols (called icons) identify each port and connector that you will use when installing your system.

2.3.3 Attaching the Ethernet Terminator or Loopback Plug

The InfoServer 1000 adapts to one of two physical Ethernet wiring schemes: ThinWire Ethernet or standard Ethernet. Your InfoServer 1000 has one of these Ethernet connections, depending upon the type ordered.

You must connect a terminator or loopback plug to the network connection so that you can run diagnostic tests to verify the internal Ethernet circuits.

_____ Note _

The system will not boot if the power-on self-test detects an Ethernet error.

The next two sections describe how to connect the Ethernet terminator or loopback plug to your specific connection. Section 2.3.3.1 describes the ThinWire Ethernet terminator. Section 2.3.3.2 describes the standard Ethernet loopback plug. Read the section for your specific Ethernet connection, then proceed to Section 2.3.4.



Figure 2–2 Standalone InfoServer 1000 Ports and Connectors

2.3.3.1 Connecting the ThinWire Ethernet Terminator

You must connect the T-connector and two Ethernet terminators (Figure 2–1) to the InfoServer 1000, so that you can run diagnostic tests to verify the Ethernet circuits.

For diagnostic purposes, the T-connector and two terminators complete the ThinWire Ethernet circuit. Later, when you decide to connect to a network, you exchange ThinWire cable sections for one or both terminators, depending on your network setup.

To connect the T-connector follow these steps:

1. Push one terminator into one side of the T-connector, then turn the terminator clockwise until it locks in place, as shown in Figure 2–3.

Figure 2–3 Attaching the Terminators to the T-Connector



- 2. Push the second terminator into the other side of the T-connector, then turn the terminator clockwise until it locks in place.
- 3. Connect the T-connector to the ThinWire Ethernet connector on the rear of the InfoServer 1000 by turning the T-connector clockwise until it locks into place (Figure 2-4).

Installing Standalone System Configurations

Figure 2–4 Connecting the T-Connector



MKV-040000331-06
2.3.3.2 Connecting the Standard Ethernet Loopback Plug

You must connect the loopback plug (Figure 2–1) to the InfoServer 1000, so that you can run diagnostic tests to verify the Ethernet circuits.

For diagnostic purposes, the loopback plug completes the standard Ethernet circuit. Connect the loopback plug to the standard Ethernet port on the back of the InfoServer 1000 by pushing it on (Figure 2-5).

Figure 2–5 Connecting the Standard Ethernet Loopback Plug



2.3.4 Connecting a Console Terminal

You do not need a locally connected terminal for normal InfoServer 1000 operation. However, Digital advises that you connect a local terminal as part of this installation procedure, to ensure that the InfoServer 1000 system is operating properly at the time of initial setup.

The recommended console terminal is a VT300 or VT400 series terminal. When you connect the terminal cable to the MMJ port on the rear of the terminal, the port must be set for DEC423 operation. Refer to your terminal documentation for details.

To connect a terminal to the server, follow these steps:

- 1. Ensure that the 10 power switch on the terminal is off (0).
- 2. Connect the terminal cable supplied with the system to the MMJ port on the rear panel of the InfoServer 1000 as shown in Figure 2-6.
- 3. Connect the other end of the console cable to the communications port on the terminal.
- 4. The terminal must be set at 9600 baud, with 8 bits, and no parity. Refer to your terminal documentation if you need information on setting up your terminal.

During normal operation, you can control the server through a LAT terminal server session (if terminal servers are available on your network). If terminal server access is not available, or if client systems have trouble accessing server services, then a locally connected terminal is needed to manage the server or to diagnose server problems. For more information, refer to the *InfoServer System Operations Guide*.



Figure 2–6 Connecting the Console Terminal Cable

MKV-040000331-09

2.3.5 Connecting the Standalone Expansion Peripherals

 \diamondsuit

If you received standalone expansion peripherals with your system, you should connect them to your InfoServer 1000 now. Refer to Chapter 6, Installing Expansion Peripherals for information on connecting expansion peripherals. After installing the expansion peripherals, return to this spot and continue with the power connections.

2.3.6 Connecting the Standalone InfoServer 1000 Power Supply and Power Cord



The H7827-BA power supply accepts power inputs in the ranges of 100 Vac to 120 Vac and 200 Vac to 240 Vac, at 50 or 60 Hz. Within these limits the power supply is self-sensing, so a voltage selection switch is not needed.

Connect the DC power cable from the power supply to the DC power input connector at the rear of the InfoServer 1000 as shown in Figure 2-7. Push the connector all the way in.

Connect the supplied AC power cord to the AC power input connector on the power supply as shown in Figure 2–7. When you are ready to apply power to the InfoServer 1000, connect the other end of the power cord to a grounded power outlet. This is the method of turning InfoServer 1000 power on and off; there is no switch.

Refer to Section 1.5 for a description of the front panel indicators.



Figure 2–7 Connecting InfoServer 1000 AC and DC Power Cords

MKV-040000331-10

2.4 Turning On and Off the System Power

To turn system power on, first turn on any standalone expansion peripherals connected to your system, then plug the InfoServer 1000 power cord into a live power socket. If a power strip is used, plug in the InfoServer 1000 and all expansion peripherals. Ensure that all expansion peripheral power switches are on, and then turn the power strip switch on.

To turn system power off, unplug the power cord or turn the power strip switch off.

Now proceed to Chapter 5, Completing the System Installation, to continue the installation.

3

Installing InfoTower System Configurations

3.1 Introduction

This chapter covers the hardware installation procedures for InfoTower packaged system configurations such as the *InfoServer Librarian*.

Installing and setting up your InfoTower mounted InfoServer 1000 system is easy. This chapter provides step-by-step instructions on how to perform the following operations:

- Choosing the right location for the system
- Unpacking, inspecting, and connecting system components
- Turning on and off the system power

Appendix A contains a brief installation checklist, without the details provided in this chapter.

3.2 Choosing the Right System Location

Use the following checklist to keep your system operating at its best:

- Carefully read all installation instructions before you turn on the power.
- Keep the temperature between 10° C and 35° C (50° F and 90° F).
- Keep the relative humidity between 10 percent and 85 percent.
- Keep the air well circulated to prevent excess heat and dust from accumulating.
- Keep your equipment away from heaters, photocopiers, direct sunlight, and abrasive particles.
- Before you set up your system, select a surface that will hold the InfoTower enclosure. Your desk or work table is a good choice.

- Keep the area clean. Do not place food or liquid on or near your equipment, and do not place your InfoServer 1000 directly on the floor. Dust and dirt will damage the system components.
- Connect your InfoServer 1000 system to a dedicated grounded circuit.

If you have several pieces of equipment that need to be plugged into a power outlet, use a grounded power strip. Many power strips come with an on/off switch and a power-surge protector.

Do not exceed the circuit power capacity.

- To avoid damaging equipment that has been moved inside from a cold environment, let the equipment warm to room temperature before you turn it on.
- When installing an InfoTower mounted system, keep the air vents clear on the front and rear of the system unit for proper ventilation. Blocking the air vents can cause the system to overheat.

3.3 Unpacking and Inspecting InfoTower System Components

CAUTION .

Because of the weight of the InfoTower system (25.5 kg (56 lb) fully loaded), two people should lift the equipment out of the shipping carton and place it on a work surface.

The shipping carton contains an accessory tray. The large carton under the accessory tray contains the system. This carton contains the InfoTower enclosure with the InfoServer 1000 system mounted in position 1 with accessories and documentation. The smaller carton, packed inside the larger carton, contains the software compact discs with licenses and instructions.

_ CAUTION _

InfoServer 1000 software is factory-loaded. The software compact discs shipped with the system are for backup purposes. Do not attempt to load software as part of the installation process. Carefully unpack all cartons. Check the contents of the cartons against the shipping list provided. Figure 3–1 shows the components and accessories shipped with the InfoServer 1000 system when mounted in an InfoTower enclosure. Expansion peripherals are shipped installed, configured, and tested in the InfoTower enclosure. Verify that the number and types of SCSI peripherals mounted in the enclosure correspond to the number and types of peripherals ordered.

	Note
Depending component for future a	on how you set up your system, you may not use all the s supplied with your system. Store any unused parts safely use.
	WARNING
For the ver mounting s	rtical installation of the InfoTower, use an enclosure stand.
	Warnung
Bei der vei Ständer fü	rtikalen Installation des InfoTowers, verwenden Sie einen r das Gehäuse.
	Avertissement
Pour une i soutenir le	nstallation veticale de l'InfoTower, utilisez un pied pour boitier.
	Quidada
Para insta montada e	lación vertical de la InfoTower usa usted una plataforma n un recinto.

Figure 3–1 System Components

Feet	
Stand	
Foot Placement	
Keys (K_ 167	
Server System Unit	
Documentation	
ACCESSORIES:	
Compact Disc Caddies	
Console Cable	
SCSI Cable (4 drive only)	
Two Terminators and One T-Connector (ThinWire Ethernet)	
SCSI Terminator	•
Loopback Plug (Standard Ethernet)	
	MKV-040000331-41

After moving the system into position, unlock and open the enclosure door. Figure 3-2 shows how a SCSI peripheral plugs into the backplane at the factory. Ensure that each peripheral is seated properly and is firmly connected to the backplane by pressing firmly on the front bezel.

Figure 3–2 Properly Seating Peripherals



MKV-040000331-12

Installing InfoTower System Configurations

Moving the System

Always repack the system in its original packing material when moving or relocating your system. This precaution will protect the hardware from damage.

3.3.1 Connecting the System Components

After unpacking, follow the instructions in Sections 3.3.2, 3.3.3, 3.3.4, 3.3.5, and 3.3.6 to connect system components.

3.3.2 Identifying the InfoTower Ports and Connectors

Figure 3-3 shows the ports, connectors, and switch on the rear of the InfoTower system. Take some time to become familiar with the ports and connectors. Symbols (called icons) identify each port and connector that you will use when installing your system.



Figure 3–3 InfoTower Ports and Connectors

Installing InfoTower System Configurations

3.3.3 Installing the SCSI Terminator

 \diamondsuit

Install the SCSI terminator on the SCSI port at the bottom rear of the InfoTower, as shown in Figure 3-4.

Figure 3–4 Installing a SCSI Terminator on an InfoTower Rear Panel



3.3.4 Attaching the Ethernet Terminator or Loopback Plug

The InfoServer 1000 adapts to one of two physical Ethernet wiring schemes: ThinWire Ethernet or standard Ethernet. Your InfoServer 1000 system has one of these Ethernet connections at the rear of the InfoTower enclosure, depending upon the type ordered.

You must connect a terminator or loopback plug to the network connection so that you can run diagnostic tests to verify the internal Ethernet circuits.

_____ Note _____

The system will not boot if the power-on self-test detects an Ethernet error.

The next two sections describe how to connect the Ethernet terminator or loopback plug to your specific connection. Section 3.3.4.1 describes the ThinWire Ethernet terminator. Section 3.3.4.2 describes the standard Ethernet loopback plug. Read the section for your specific Ethernet connection, then proceed to Section 3.3.5.

Installing InfoTower System Configurations

3.3.4.1 Connecting the ThinWire Ethernet Terminator

You must connect the T-connector and two Ethernet terminators (Figure 3-1) to the system, so that you can run diagnostic tests to verify the Ethernet circuits.

For diagnostic purposes, the T-connector and two terminators complete the ThinWire Ethernet circuit. Later, when you decide to connect to a network, you exchange ThinWire cable sections for one or both terminators, depending on your network setup.

To connect the T-connector follow these steps:

1. Push one terminator into one side of the T-connector, then turn the terminator clockwise until it locks in place, as shown in Figure 3-5.

Figure 3–5 Attaching the Terminators to the T-Connector



- 2. Push the second terminator into the other side of the T-connector, then turn the terminator clockwise until it locks in place.
- 3. Connect the T-connector to the ThinWire Ethernet connector on the rear of the InfoServer 1000 by turning the T-connector clockwise until it locks into place (Figure 3-6).



Figure 3–6 Connecting the T-Connector

Installing InfoTower System Configurations

3.3.4.2 Connecting the Standard Ethernet Loopback Plug

You must connect the loopback plug (Figure 3–1) to the system, so that you can run diagnostic tests to verify the Ethernet circuits.

For diagnostic purposes, the loopback plug completes the standard Ethernet circuit. Connect the loopback plug to the standard Ethernet port on the back of the InfoTower enclosure by pushing it on (Figure 3-7).

Figure 3–7 Connecting the Standard Ethernet Loopback Plug



3.3.5 Connecting a Console Terminal

You do not need a locally connected terminal for normal InfoServer 1000 operation. However, Digital advises that you connect a local terminal as part of this installation procedure, to ensure that the InfoServer 1000 system is operating properly at the time of initial setup.

The recommended console terminal is a VT300 or VT400 series terminal. When you connect the terminal cable to the MMJ port on the rear of the terminal, the port must be set for DEC423 operation. Refer to your terminal documentation for details.

To connect a terminal to the server, follow these steps:

- 1. Ensure that the 10 power switch on the terminal is off (0).
- 2. Connect the terminal cable supplied with the system to the MMJ port on the rear panel of the InfoServer 1000 as shown in Figure 3-8.
- 3. Connect the other end of the console cable to the communications port on the terminal.
- 4. The terminal must be set at 9600 baud, with 8 bits, and no parity. Refer to your terminal documentation if you need information on setting up your terminal.

During normal operation, you can control the server through a LAT terminal server session (if terminal servers are available on your network). If terminal server access is not available, or if client systems have trouble accessing server services, then a locally connected terminal is needed to manage the server or to diagnose server problems. For more information, refer to the *InfoServer System Operations Guide*.

Installing InfoTower System Configurations

Figure 3–8 Connecting the Console Terminal Cable



3.3.6 Connecting System Power to InfoTower Configurations

InfoServer 1000 system configurations (CPU and peripherals) housed in the InfoTower enclosure draw their power from a single internal power supply. The InfoTower enclosure accepts power inputs in the ranges of either 100 Vac to 120 Vac or 200 Vac to 240 Vac, at 50 or 60 Hz. Perform the following steps to connect power (Figure 3-9):

- 1. Ensure that the 10 power switch on the rear panel of the system unit is off (O).
- 2. Locate and remove the label over the power supply connector.
- 3. Check the power selection switch. Ensure that it matches your power source voltage range. If necessary, flip the switch from one side to the other.
- 4. Connect the supplied AC power cord to the AC power input connector at the rear of the InfoTower enclosure.
- 5. Connect the other end of the power cord to a grounded power outlet.

Refer to Section 1.5 for a description of the front panel indicators.

Figure 3–9 Connecting the InfoTower Enclosure Power Cord





3.4 Turning On and Off the System Power

١

To turn system power on, set the $\boxed{10}$ power switch on the rear of the enclosure to the on (1) position.

To turn system power off, set the $\boxed{10}$ power switch on the rear of the enclosure to the off (0) position.

Now proceed to Chapter 5, Completing the System Installation, to continue the installation.

4

Installing StorageWorks System Configurations

4.1 Introduction

This chapter describes the hardware installation procedures for the following StorageWorks InfoServer SW1000 system configurations:

- InfoServer SW1000-a half-height device mounted in the top of a 5.25-inch StorageWorks Building Block (SBB).
- Factory-configured BA350-KA pedestals containing InfoServer SW1000.

Installing and setting up your StorageWorks InfoServer SW1000 system is easy. This chapter provides step-by-step instructions on how to perform the following operations:

- Choosing the right location for the system
- Unpacking, inspecting, and connecting system components
- Turning on and off the system power

Appendix A contains a brief installation checklist, without the details provided in this chapter.

4.2 Choosing the Right System Location

Use the following checklist to keep your system operating at its best:

- Carefully read all installation instructions before you turn on the power.
- Keep the temperature between 18° C and 24° C (50° F and 90° F).
- Keep the relative humidity between 40 percent and 60 percent.
- Keep the air well circulated to prevent excess heat and dust from accumulating.
- Keep your equipment away from heaters, photocopiers, direct sunlight, and abrasive particles.

- Before you set up your system, select a location that accomodates the StorageWorks InfoServer SW1000, associated StorageWorks Shelf and expansion peripherals. Beside your desk or work table is a good choice.
- Keep the area clean. Do not place food or liquid on or near your equipment. Dust and dirt will damage the system components.
- Connect your StorageWorks Pedestal to a dedicated grounded circuit.

If you have several pieces of equipment that need to be plugged into a power outlet, use a grounded power strip. Many power strips come with an on/off switch and a power-surge protector.

Do not exceed the circuit power capacity.

• To avoid damaging equipment that has been moved inside from a cold environment, let the equipment warm to room temperature before you turn it on.

4.3 Unpacking and Inspecting the StorageWorks System Components

Figure 4-1 shows the components shipped with the StorageWorks pedestal containing a factory-installed InfoServer SW1000 system or the InfoServer SW1000 SBB. Check the contents of the cartons against the shipping list provided. Expansion storage devices are shipped in separate cartons (refer to Section 6.3), unless the entire StorageWorks system has come from the factory, configured to order.



Figure 4–1 InfoServer SW1000 System Components

MK2709-1

The StorageWorks InfoServer SW1000 system is shipped in one carton. This carton contains the InfoServer SW1000 hardware with accessories and documentation. In addition, packed inside the larger carton, other cartons containing the software compact discs with licenses and instructions may be found. StorageWorks expansion peripherals may be shipped in separate cartons.

_ CAUTION ____

InfoServer SW1000 software is factory-loaded. The software compact discs shipped with the system are for backup purposes. Do not attempt to load software as part of the installation process.

_____ Note __

Depending on how you set up your system, you may not use all the components supplied with your system. Store any unused parts safely for future use.

Moving the System

Always repack the InfoServer SW1000 in its original packing material when you move or relocate your system. This precaution protects the hardware from damage.

4.4 Installing the InfoServer SW1000

If you have ordered an InfoServer SW1000 SBB, use the instructions in Sections 4.4.1, 4.4.2, 4.4.3, 4.4.5, and 4.4.6, to install the unit.

If you have ordered a factory-configured StorageWorks pedestal containing the InfoServer SW1000, use sections 4.4.3, 4.4.4, 4.4.6 for installation.

4.4.1 Identifying InfoServer SW1000 Ports and Connectors

Figure 4-2 shows the console port and the Ethernet connector on the front of the InfoServer SW1000 system. Take some time to become familiar with the ports and connectors. Symbols (called icons) identify each port and connector that you will use when installing your system.



Figure 4–2 InfoServer SW1000 Ports and Connectors

4.4.2 Determine SCSI Bus Configuration and Termination

The InfoServer SW1000 comes with one SCSI port and can support up to 7 SCSI devices. This SBB has a device address factory set to an ID of 7. The physical switch on the rear of the InfoServer SW1000 SBB is nonfunctional for this product.

SCSI bus terminators should be disabled or removed on all SBB storage devices. The InfoServer SW1000 provides termination internally.

InfoServer SW1000 must be placed in slot 0 of your StorageWorks pedestal, establishing the electrical and logical beginning of the SCSI bus. Connector JA1 should not be used. InfoServer SW1000 configuration rules are described as follows:

Single Shelf, Single Bus

The single shelf, single bus configuration dedicates the StorageWorks pedestal to the InfoServer SW1000 allowing network-shareable storage for up to seven peripherals. See Figure 4–3 for a display of SCSI bus lengths for this configuration of InfoServer SW1000

- JA1 contains no terminator or expansion cable.
- InfoServer SW1000 is set to ID 7 (factory configured).
- InfoServer SW1000 is installed at slot 0 (factory configured).
- External expansion peripherals can be supported through JB1 if the terminator behind slot 1 is removed and the SCSI bus length restrictions are observed.



Figure 4–3 Single Shelf, Single Bus Configuration

CXO-4110A-MC

Table 4–1 defines the type and maximum length of the StorageWorks SCSI bus as measured between the two bus terminators, when used with the InfoServer SW1000.

Table 4-	I SCSI	Bus I	Parameters
----------	--------	--------------	------------

Bus Type	Transfer Rate	Meters	Feet
8-bit, single-ended	5 MB/s	6	19.7
) <u> </u>	
The InfoSomies SI	W1000 doos not nroe		ing the Storege Warks

The InfoServer SW1000 does not process or utilize the StorageWorks pedestal Shelf_OK signal.

Single Shelf, Two Buses

A typical application for a split bus configuration may have the InfoServer SW1000 providing the data center with network-shareable storage resources while allowing a host file server or mainframe access to non-shared storage contained in the same StorageWorks pedestal. See Figure 4-4 for a display of SCSI bus lengths for this configuration of InfoServer SW1000. Table 4-1 defines the type and maximum length of the StorageWorks SCSI bus as measured between the two bus terminators, when used with the InfoServer SW1000.

- JA1 contains no terminator or expansion cable.
- InfoServer SW1000 is set to ID 7 (factory configured).
- Companion devices contained in the InfoServer SW1000 5.25-inch SBB should utilize the right-most (viewed from the rear of the SBB) backplane connector position to ensure both devices are on the same SCSI bus (factory configured).
- InfoServer SW1000 is installed at slot 0 (factory configured).
- No external expansion peripherals can be supported through JA1.
- Refer to StorageWorks Family User's Guide for the JB1 configuration rules.



Figure 4-4 Single Shelf, Two Buses Configuration

CXO-4111A-MC

Double Shelf, Single Bus

InfoServer SW1000 supports up to seven devices. A single StorageWorks pedestal may not allow the maximum number of devices to be installed. To fully utilize the InfoServer SW1000 SCSI bus, expansion storage can be added in a second StorageWorks pedestal and the two units can be tied together. See Figure 4-5 for a display of SCSI bus length for this configuration of InfoServer SW1000. Table 4-1 defines the type and maximum length of the StorageWorks SCSI bus as measured between the two bus terminators, when used with the InfoServer SW1000.

- JA1 contains no terminator or expansion cable in Shelf 1.
- InfoServer SW1000 is set to ID 7.
- InfoServer SW1000 is installed at slot 0 in Shelf 1.
- External expansion peripherals can be supported through JB1 of Shelf 2 if the terminator behind slot 1, Shelf 2 is removed and the SCSI bus length restrictions are observed.



Figure 4–5 Double Shelf, Single Bus Configuration

CXO-4112A-MC

4.4.3 Insert InfoServer SW1000 into StorageWorks Pedestal

You can insert the InfoServer SW1000 into your existing StorageWorks pedestal. Figure 4-6 shows how a InfoServer SW1000 SBB can be directly inserted into the pedestal. The SCSI bus plugs into the backplane of the StorageWorks pedestal.

Note .

Once the InfoServer SW1000 is installed, the StorageWorks pedestal SCSI bus should not be connected to other file servers, otherwise, SCSI bus conflicts may result.



Figure 4–6 Inserting the SBB into StorageWorks Pedestal

The InfoServer SW1000 takes up the top three slots in a StorageWorks pedestal. To add the InfoServer SW1000 to your existing StorageWorks pedestal, insert the SBB into the top guide slots and push it in until it is fully seated and the mounting tabs engage the shelf. Ensure that the InfoServer SW1000 is seated properly and is firmly connected to the backplane by pressing firmly on the front bezel.

4.4.3.1 Installing a Second 5.25-Inch Half-Height Device

The StorageWorks InfoServer SW1000 SBB has space to install a second halfheight device. It can be another 5.25-inch device, such as, a CD-ROM reader or tape drive. When two half-height devices are installed in one 5.25-inch SBB, the second device is on the right. The InfoServer is always installed as the upper device, or on the left on a StorageWorks shelf. Refer to the StorageWorks Family 5 1/4-Inch SBB Half-Height Device Installation Guide, The StorageWorks Family User Guide, and StorageWorks Family Configuration Guide for installation guidance.

Companion devices contained in the InfoServer SW1000 5.25-inch SBB should utilize the right-most (viewed from the rear of the SBB) backplane connector position to ensure both devices are on the same SCSI bus.

4.4.4 Installing a Pre-Configured InfoServer SW1000

Figure 4–7 shows a configuration of InfoServer SW1000 and CD-ROM devices in the StorageWorks pedestal. This is representative of a typical configuration. Additional combinations of device options may be configured to fit your needs.

Figure 4-7 StorageWorks Pedestal with InfoServer SW1000



Follow the unpacking and identifying components directions, then continue with Sections 4.4.6 through 4.5.

4.4.5 Connecting a Console Terminal

You do not need a locally connected terminal for normal InfoServer SW1000 operation. However, Digital advises that you connect a local terminal as part of this installation procedure, to ensure that the InfoServer SW1000 system is operating properly at the time of initial setup.

The recommended console terminal is a VT300 or VT400 series terminal. When you connect the terminal cable to the MMJ port on the front of the terminal, the port must be set for DEC423 operation. Refer to your terminal documentation for details.

To connect a terminal to the server, follow these steps:

- 1. Ensure that the 10 power switch on the terminal is off (0).
- 2. Connect the terminal cable supplied with the system to the MMJ port on the front panel of the InfoServer SW1000 as shown in Figure 4-8.
- 3. Connect the other end of the console cable to the communications port on the terminal.
- 4. The terminal must be set at 9600 baud, with 8 bits, and no parity. Refer to your terminal documentation if you need information on setting up your terminal.

During normal operation, you can control the server through a LAT terminal server session (if terminal servers are available on your network). If terminal server access is not available, or if client systems have trouble accessing server services, then a locally connected terminal is needed to manage the server or to diagnose server problems. For more information, refer to the *InfoServer System Operations Guide*.


Figure 4–8 Connecting the Console Terminal Cable

IVIK2/09-3

4.4.6 Attaching the Ethernet Terminator

The InfoServer SW1000 adapts to the ThinWire Ethernet wiring scheme.

You must connect a terminator to the network connection so that you can run diagnostic tests to verify the internal Ethernet circuits.

____ Note ____

The system will not boot if the power-on self-test detects an Ethernet error.

The 4.4.6.1 describes how to connect the Ethernet terminator to your specific connection and describes the ThinWire Ethernet terminator.

Installing StorageWorks System Configurations

4.4.6.1 Connecting the ThinWire Ethernet Terminator

You must connect the T-connector and two Ethernet terminators (Figure 4-1) to the InfoServer SW1000, so that you can run diagnostic tests to verify the Ethernet circuits.

For diagnostic purposes, the T-connector and two terminators complete the ThinWire Ethernet circuit. Later, when you decide to connect to a network, you exchange ThinWire cable sections for one or both terminators, depending on your network setup.

To connect the T-connector follow these steps:

1. Push one terminator into one side of the T-connector, then turn the terminator clockwise until it locks in place, as shown in Figure 4-9.

Figure 4–9 Attaching the Terminators to the T-Connector



- 2. Push the second terminator into the other side of the T-connector, then turn the terminator clockwise until it locks in place.
- 3. Connect the T-connector to the ThinWire Ethernet connector on the front of the InfoServer SW1000 by turning the T-connector clockwise until it locks into place (Figure 4-10).



Figure 4–10 Connecting the T-Connector

4.5 Turning On and Off the System Power

To turn system power on, first turn on any expansion peripherals connected externally to the StorageWorks shelf. Then plug the StorageWorks power cord into a live power socket.

To turn on system power follow these steps:

- 1. Ensure that the [0] power switch on the StorageWorks Pedestal is off (0).
- 2. Plug in the StorageWorks Pedestal.
- 3. Turn on the console terminal (if any).
- 4. Turn on the external expansion peripherals (if any).
- 5. Turn on the InfoServer SW1000 pedestal by pressing the 10 power switch to the ON (1) position.

Installing StorageWorks System Configurations

To turn the system power off, press the $\boxed{10}$ switch to the OFF (0) position. Now proceed to Chapter 5, Completing the System Installation, to continue the installation.

Completing the System Installation

5.1 Introduction

The information in this chapter concludes the system installation. Starting and booting your system involves only turning the InfoServer 1000 system power on. The InfoServer SW1000 powers on with the rest of the StorageWorks configuration. This chapter explains what the console terminal displays when InfoServer 1000 power is turned on. It also directs you to connect the system to a network after starting the system initially.

5.2 Starting the System

After installing your system and attaching a network terminator or loopback plug, the system may be started as a standalone configuration. To start your system, follow these steps:

- 1. Turn on any standalone expansion peripherals connected to your system.
- 2. Turn on the console terminal connected to the MMJ port. Refer to your terminal documentation. Note that a console terminal is required when the server is first installed to ensure proper startup and cabling.

Note .

To view the system power-on self-test display, you must allow time for the console terminal to warm up and complete its power-on sequence.

3. Turn on the InfoServer 1000 or the StorageWorks pedestal.

The initial power-on display appears on the console terminal in a few seconds. If you see a display similar to the following, your system has passed all power-on self-tests:

The system now boots the operating system from the internal memory. If the boot is successful, the console screen displays a message similar to the following:

```
83 BOOT SYS
-DKA200
-DKA100
-FLASH
InfoServer 1000 V3.1
Copyright (c) 1990, 1991, 1992, 1993 Digital Equipment Corp.
%ESS-I-CONFIGDEV, Device configuration complete. 2 devices found
InfoServer 1000 V3.1
```

4. Enter the system password (default is ESS) when the system displays the Enter Password: prompt.

If you do not see the power-on display or the server fails to bootstrap, turn the power off and review each installation step. Repeat the power-on procedure. If the problem continues, refer to Section 8.12.

5.3 Connecting the System to a Network

After you verify that the InfoServer 1000 system passes its power-on self-test sequence, you must connect the InfoServer 1000 system to the network to make the system available to clients. To connect the InfoServer 1000 system to a network, follow these steps:

- 1. Turn the InfoServer 1000 or the StorageWorks pedestal power off.
- 2. Read and perform the steps in Chapter 7, Connecting to a Network, to become familiar with how to install your network hardware.
- 3. After making the network connection, turn the InfoServer 1000 power back on. The server should pass the power-on self-test and reboot successfully.
- 4. Refer to the InfoServer System Operations Guide for information on how to establish a server management session to set up and control the server.

5.4 Booting the System

The InfoServer 1000 operating system starts as part of the bootstrap process when power is first applied. The following sections explain the automatic and manual processes for starting the InfoServer 1000. Use the manual process only if the automatic process fails.

5.4.1 Booting the System Automatically Without a Console

The InfoServer 1000 system's bootstrap program and operating software are loaded into the system at the factory. The InfoServer 1000 attempts to boot automatically from the internal memory when power is applied.

5.4.2 Booting the System with a Console (If Necessary)

To boot your system with a console terminal, follow these steps:

- 1. Turn on any standalone expansion peripherals connected to your InfoServer 1000.
- 2. Turn on the console terminal connected to the MMJ port.

Note _

Allow time for the console terminal to complete its power-on sequence. Otherwise, you will not be able to view the system power-on self-test display.

3. Turn on the InfoServer 1000 or the StorageWorks pedestal.

The initial power-on display appears on the console terminal in a few seconds. If you see a display similar to the following, your system has passed all power-on self-tests, and the terminal's keyboard language has been set:

The system now boots the operating system from the internal memory. If the boot is successful, the console screen displays a message similar to the following:

```
83 BOOT SYS
-DKA200
-DKA100
-FLASH
InfoServer 1000 V3.1
Copyright (c) 1990, 1991, 1992, 1993 Digital Equipment Corp.
%ESS-I-CONFIGDEV, Device configuration complete. 2 devices found
InfoServer 1000 V3.1
Enter Password:
```

4. Enter the system password (default is ESS) when the system displays the Enter Password: prompt.

Installing Expansion Peripherals

6.1 Introduction

This chapter provides information on installing and using external standalone add-on storage peripherals. For information on installing and using addon storage peripherals for the InfoTower cabinet, refer to the InfoTower Installation and Owner's Guide. For information on installing and using addon storage peripherals in a StorageWorks pedestal refer to the StorageWorks Family Configuration Guide.

An InfoServer 1000 system can support up to seven (7) standalone expansion peripherals on the SCSI bus. Digital qualifies the standalone expansion peripherals recommended for use with the InfoServer 1000 system, to ensure that they meet system requirements.

The chapter explains how to perform the following tasks:

- Connect one or two expansion peripherals of the same type to your InfoServer 1000.
- Verify the SCSI ID default (factory) setting.
- Reset the SCSI ID number.

The chapter provides specific information for the CD-ROM expansion drive. For installation and operation information pertaining to other SCSI expansion peripherals, refer to the documentation shipped with the peripheral.

6.2 Guidelines for Connecting Expansion Peripherals

Use the following guidelines when connecting expansion peripherals:

- Connect any combination of external expansion peripherals, as long as you do not connect more than seven (7) expansion peripherals.
- Attach the 50-pin SCSI terminator that comes with your expansion peripheral to an unused bottom SCSI connector on the back of the expansion peripheral. This must be done to maintain signal integrity.
- Plug all expansion peripherals into the same grounded power strip or electrical outlet.

CAUTION .

Some expansion peripherals are restricted to either 110 Vac or 220 Vac operation. Ensure that the voltage requirement of any expansion peripheral is compatible with the supply voltage.

Some expansion peripherals accept power inputs in the ranges of 100 Vac to 120 Vac and 200 Vac to 240 Vac, at 50 or 60 Hz. Within these limits the power supply is self-sensing, and a voltage selection switch is not necessary.

• Turn on all expansion peripherals before you turn on your InfoServer 1000. This step ensures that the device in each expansion peripheral is ready for use, and that the system firmware includes the device in its configuration.

6.3 Unpacking an Expansion Peripheral

SCSI expansion peripherals in external, standalone enclosures usually come with the following accessories:

- One 50-pin to 50-pin cable (PN BC19J-1E). Use this cable to connect the expansion peripheral to other expansion peripherals.
- One 68-pin to 50-pin cable (PN BC56H-03). This cable is not used on the InfoServer 1000.
- One 50-pin SCSI terminator.
- One power cord.

Before you set up your expansion peripheral, ensure that you have all these parts. Check the contents of the expansion peripheral against the shipping list.

6.4 Connecting One Expansion Peripheral

To connect a single expansion peripheral to your InfoServer 1000, follow these steps:

- 1. Turn system power off.
- 2. Turn the expansion peripheral so that the back is facing you.
- 3. Set the SCSI ID number.

Each device normally ships with the SCSI ID number set to 0 if the ID is switch selectable. However, if the SCSI ID is jumper selectable, the SCSI ID is usually set to 7. Refer to the information provided with the expansion peripheral to verify this setting.

The standalone drive has a rear-panel push button that spins a wheel and advances a number. To set the device ID number, press the button until the desired number appears.

4. Connect the BC09D-01 50-pin expansion peripheral cable as follows:

Connect the high-density 50-pin end of the cable to the SCSI port on the rear of the InfoServer 1000.

Connect the opposite 50-pin end of the cable to the top SCSI port on the rear of the expansion peripheral. Snap the spring lock loops in place.

5. Connect a 50-pin SCSI terminator to the lower port on the expansion peripheral.

_ CAUTION _

Check the voltage requirement of the expansion peripheral before applying power.

6. Connect the power cord to the power receptacle on the expansion peripheral, then plug the other end into a grounded power outlet.

Figure 6-1 shows how to connect one CD-ROM expansion peripheral to the InfoServer 1000. You can stack the InfoServer 1000 on top of the expansion peripheral or place the two enclosures side by side.

Figure 6–1 Connecting One CD–ROM Expansion Peripheral



.

6.5 Connecting Two Expansion Peripherals

A second expansion peripheral can be added later to link the two expansion peripherals together in series. You can stack expansion peripherals or place them side by side.

Note _____

If you have more than one of the same type of device attached to your system, you must change the SCSI ID on the second device to an open ID number on the SCSI bus (refer to Appendix C).

To link two expansion peripherals, follow these steps:

- 1. Ensure that the <u>IO</u> on/off switches on all expansion peripherals are set to off (O), then remove power from the InfoServer 1000.
- 2. Remove the 50-pin SCSI terminator on the first expansion peripheral. Store the terminator for future use.
- 3. Connect one end of the BC19J-1E 50-pin to 50-pin connector cable to the upper connector on the second expansion peripheral.
- 4. Connect the other end of the cable to the lower connector on the first expansion peripheral.
- 5. Attach a 50-pin terminator to the unused SCSI port on the second expansion peripheral.
- 6. Connect the power cord of the second expansion peripheral to its power receptacle, then plug the other end into a grounded power outlet. Each expansion peripheral needs its own source of power.

Figure 6-2 shows how to connect two CD-ROM expansion peripherals to the InfoServer 1000 in series.



Figure 6–2 Connecting Two CD–ROM Expansion Peripherals

6.6 CD-ROM Compact Disc Expansion Peripheral

The CD-ROM is a read-only storage device that reads data from removable compact discs. Refer to Appendix B for information on the handling and storage of compact discs.

If you are installing a CD-ROM expansion peripheral, refer to the beginning of this chapter for unpacking and installation instructions.

6.6.1 Verifying the SCSI ID Number on the CD–ROM Expansion Peripheral

Figure 6-3 shows the rear panel of the CD-ROM expansion peripheral, which ships with the SCSI ID number set to 0. The number is displayed in a window beneath the ID selector switch.



Figure 6–3 CD–ROM Rear Panel

Installing Expansion Peripherals

6.6.2 Determining Available SCSI ID's

If you add peripherals to your system, you must set the ID number on the subsequent peripheral to an open ID - one that is not in use. To reset the number, follow instructions in step 3 of Section 6.4. In the following example, IDs 0, 4, 5, and 7 are open IDs.

To determine available ID numbers on the SCSI bus on your InfoServer 1000 system, enter SHOW DEVICE at the console prompt (>>>), then press Return. For instructions on how to enter console mode, refer to Section 8.4. Your system configuration appears on the screen. Look for a display similar to the following:

>>> SHOW DEV	/ICE						
VMS/VMB	ADDR	DEVTYPE	NUMBYTES	RM/FX	WP	DEVNAM	REV
 7230			mh i sti se				
DKA100	A/1/0	DISK	121.64MB	FX		RZ23L	2528
DKA200	A/2/0	DISK	593.49MB	RM	WP	RRD42	1.21
DKA300	A/3/0	DISK	600 MB	RM	WP	RRD43	0064
HostID	A/6	INITR	•	6	A	9	A
	8	0	9	0	U	U	0



2 Controller/SCSI ID/logical number

In this example: SCSI ID 0 — Open SCSI ID 1 - RZ23L hard-disk drive SCSI ID 2 - RRD42 CD-ROM drive SCSI ID 3 - RRD43 CD-ROM drive SCSI ID 4 — Open SCSI ID 5 - Open SCSI ID 6 — Internal SCSI controller (open for SW1000) SCSI ID 7 — Open (Internal SCSI controller for SW1000)



6 Generic device type Device capacity



5 Removable/fixed media

6 Write protected

Specific device type

8 Device firmware revision level

_____ Note _____

InfoServer SW1000 systems utilize a host ID of 7.

6.7 Preventive Maintenance

The CD-ROM drive does not require periodic cleaning. To maintain maximum performance, ensure that the drive's environment is free from contamination.

If errors occur, clean the disc with a clean, lint-free cloth. Wipe the disc from the center out. Do not use solvents, such as benzene, thinner, commercial cleaners, or antistatic sprays.

Follow these guidelines to care for the caddy and the disc:

- Do not drop the disc or caddy.
- The caddy's shutter automatically opens when it is inserted into the drive. Do not open the shutter manually and touch the disc.
- When transporting the the standalone CD-ROM drive or the InfoTower, remove the caddy from the drive.
- Do not put the disc and caddy in a location subject to any of the following:

Dust Direct sunlight High temperature High humidity

• When a disc is dirty, clean it.

• v

7

Connecting to a Network

7.1 Introduction

This chapter explains how to:

- Connect the InfoServer 1000 to a ThinWire Ethernet or standard Ethernet network.
- Test the system with a T-connector or loopback connector.
- Connect several systems together to create a simple ThinWire Ethernet workgroup.

To form a workgroup, connect the systems together with cables and connectors.

Connecting a system to a ThinWire Ethernet network or a standard Ethernet network takes only a few minutes for an experienced user and slightly longer for a new user.

The illustrations in this chapter show Ethernet connections to the standalone InfoServer 1000. The connections are identical when working with the InfoServer 1000 mounted in an InfoTower cabinet or an InfoServer SW1000.

If you are connecting your InfoServer 1000 or InfoServer SW1000 to a ThinWire Ethernet network, proceed to Section 7.2.

If you are connecting your InfoServer 1000 to a standard Ethernet network, proceed to Section 7.4.

7.2 Connecting to a ThinWire Ethernet Network

The following sections explain how to connect the InfoServer 1000 system to a ThinWire Ethernet network.

7.2.1 Verifying Your ThinWire Ethernet Network Adapter

To verify the ThinWire Ethernet adapter, it must be correctly terminated. Be sure that the ThinWire Ethernet T-connector and terminators are attached to your system. You also need a console terminal.

Follow these steps:

- 1. Force the system into console mode as follows:
 - a. Turn the InfoServer 1000 power off.
 - b. Disconnect the network T-connector.
 - c. Turn the InfoServer 1000 power on.

When the system runs its power-on self-test diagnostics, the network test will fail and the system will go into console mode. A display similar to the following appears:

- d. Reconnect the network T-connector.
- 2. Enter the following command (Test NI):

>>> T NI Return

The system responds with another display similar to the following:

>>> T NI | | ### OK >>> If the system responds with a display similar to the following, there is a problem with the InfoServer 1000 network adapter:

```
>>>T NI
| |
###
?? 001 6 NI 0172
84 FAIL
>>>
```

7.2.2 Connecting Your System to a ThinWire Ethernet Cable

To connect the server system to ThinWire Ethernet cable, follow these steps:

- 1. Remove power from the system.
- 2. Attach a ThinWire cable to one side of the T-connector for the InfoServer 1000. Figure 7-1 shows the position of the T-connector.
- 3. If your system is the first or last system in a single ThinWire segment, leave a terminator attached to one side of the T-connector.

A segment is a length of cable made up of one or more cable sections connected with **barrel connectors** or T-connectors.

4. If your system *is not* the first or last system and you want to connect another system to the ThinWire cable, add another section of cable to the other side of the T-connector.

Ensure that the T-connector is attached correctly to the InfoServer 1000 or InfoServer SW1000. Push the T-connector onto the connector, then turn the T-connector clockwise until it locks in place.

Connecting to a Network

Figure 7–1 Attaching the T-Connector with Cable Segment to the BNC Connector



MKV-040000331-24

7.3 Creating a ThinWire Ethernet Workgroup

A workgroup is a set of systems connected serially on the same ThinWire segment. This segment can work as a standalone network or can be connected to a larger network.

The serial configuration lowers the cost of wiring. However, there is a risk that someone may unplug a system and make the network connections inoperative between systems on the ThinWire segment.

'.3.1 Connecting Your System to an OPEN DECconnect Faceplate

If your office is wired with Digital's OPEN DECconnect products, then connect your system to the OPEN DECconnect faceplate in your office. A **faceplate** is a wall receptacle that provides a network connection for your system.

OPEN DECconnect cabling components support network configurations ranging from simple standalone ThinWire networks to multimedia local area networks (LANs) that connect PCs, DECstations, VAXstations, and VAX systems. No matter how large or how small your network, the most visible OPEN DECconnect component in your office is the faceplate.

The faceplate has one of the following connectors:

- A ThinWire BNC connector that allows several systems, including InfoServer 1000 systems, to be serially connected to one another on the office side of the faceplate.
- A ThinWire Ethernet serial connector that allows several faceplates to be serially connected behind the wall to a single segment of ThinWire cable. One system can be attached to each faceplate.

Your network coordinator knows which type of OPEN DECconnect faceplate you have in your office.

To connect your system, attach one end of the ThinWire cable to the office side of the OPEN DECconnect faceplate. Attach the other end of the ThinWire cable to the ThinWire connector on the back of your system. Figure 7-2 shows these connections.

Figure 7–2 Connecting to the OPEN DECconnect Faceplate



MKV-040000331-25

7.3.2 Troubleshooting the ThinWire Segment

If you find problems during the ThinWire Ethernet installation, use the following checklist to ensure that all connections are correct:

- 1. Check that all connections are secure on the ThinWire Ethernet segment.
- 2. Check that there is a T-connector between the connector on the ThinWire cable and the InfoServer 1000.
- 3. Check that there is no ThinWire segment between the system and the T-connector. The T-connector must attach directly to the system.
- 4. Check that there are no branches off a ThinWire segment except through a DEMPR multiport repeater.

A DEMPR provides eight (8) ThinWire Ethernet connections from a single standard Ethernet connection.

- 5. Check that a ThinWire segment is not looped from one port on the DEMPR to another port on the DEMPR.
- 6. Check that the InfoServer 1000's T-connector has not been disconnected from an operating ThinWire Ethernet segment.

_____ CAUTION ____

If you need to remove a system from an active ThinWire segment, disconnect the center of the T-connector from the system, as shown in Figure 7–3.

7. Check that you have not disconnected a terminator from an active ThinWire segment. This will disrupt network operation.

After you have checked all ThinWire Ethernet connections, retest the Ethernet as directed in Section 7.2.1.

Connecting to a Network

Figure 7–3 Disconnecting the T-Connector



MKV-040000331-26

7.4 Connecting to a Standard Ethernet Network

The following sections explain how to connect the InfoServer 1000 system to a standard Ethernet network.

7.4.1 Verifying Your Standard Ethernet Network Adapter

To verify the standard Ethernet adapter, you must attach the 15-pin standard Ethernet loopback plug to your InfoServer 1000.

Follow these steps:

- 1. Force the system into console mode as follows:
 - a. Turn the InfoServer 1000 power off.
 - b. Disconnect the network loopback plug at the rear of the enclosure.
 - c. Turn the InfoServer 1000 power on.

When the system runs its power-on self-test diagnostics, the network test will fail and the system will go into console mode. A display similar to the following appears:

- d. Reconnect the loopback plug. The indicator on the loopback plug should be on.
- 2. Enter the following command (Test NI):

>>> T NI Return

The system responds with another display similar to the following:

>>> T NI | | ### OK >>> If the system responds with a display similar to the following, there is a problem with the InfoServer 1000 network adapter:

```
>>>T NI
| |
###
?? 001 6 NI 0172
84 FAIL
>>>
```

3. After successful network verification, remove the loopback plug and store it for future diagnostic testing. Connect your system to a standard Ethernet cable, as described in the next section.

If you have problems, refer to Section 7.4.3.

7.4.2 Connecting a Transceiver Cable to Your System

A transceiver cable is the physical connection between a standard Ethernet network interface and an InfoServer 1000 system. You use a transceiver cable to connect your system to an H4005 transceiver or DELNI unit, which connects to the standard Thickwire Ethernet cable. To connect a transceiver cable to your system:

- 1. Remove power from the InfoServer 1000 system.
- 2. Connect the 15-pin connector on the transceiver cable to the standard Ethernet connector on the rear of the InfoServer 1000.
- 3. Use a screwdriver to move the connector's sliding lock and make the standard Ethernet connection secure as shown in Figure 7-4.





MKV-040000331-27

7.4.3 Troubleshooting the Standard Ethernet Connection

If you have problems verifying the standard Ethernet connection, follow these steps:

1. During standalone verification, ensure that the loopback plug is securely connected to the system.

If you are connected to an active Ethernet segment, you can test the system without the loopback plug.

- 2. Turn your system off, then turn it on again. The system verifies the Ethernet connection during the power-on sequence.
- 3. After you check all Ethernet connections, retest the Ethernet adapter as directed in Section 7.4.1.

.

.

Solving Problems

8.1 Introduction

This chapter provides information on troubleshooting problems with the InfoServer 1000 system. Topics include:

- Checking cable connections
- Using the troubleshooting table
- Entering the console mode
- Interpreting power-on error messages
- Running self-tests
- Displaying system configuration information
- Displaying device information
- Running test utilities for expansion peripherals
- Restarting the system after running tests
- Reestablishing console parameters
- Recovering from bootstrap failures
- Service information
- Command summary

8.2 Checking the Cable Connections

If a problem occurs, first make sure that all power (ac and dc), network, and SCSI device cables are connected correctly. Follow these steps to check the cables:

- 1. Turn off the console terminal.
- 2. Turn off all expansion peripherals.

Solving Problems

- 3. Turn off the InfoServer 1000 or the StorageWorks pedestal.
- 4. With all equipment turned off, check that the following cables are correctly connected at both ends:
 - Console cable (linking the console terminal or LAT port to the InfoServer 1000)
 - Terminal power cord
 - InfoServer 1000, InfoTower, InfoServer SW1000 power cord(s)
 - SCSI cables from standalone InfoServer 1000 to standalone expansion peripherals
 - Standalone peripheral power cords

Now, apply power to your equipment in the following order:

- 1. Turn on the standalone expansion peripherals (if any).
- 2. Turn on the console terminal.
- 3. Turn on the InfoServer 1000, the InfoTower, or the StorageWorks pedestal.

The power-on self-test display should appear and the InfoServer 1000 system should boot. The console display should look similar to the following example:

After the last line of the display, the InfoServer 1000 operating system kernel is running and the System OK indicator flashes on and off once/second. If the display does not resemble this example, or if any question marks appear in the display, refer to Section 8.3.

8.3 Using the Troubleshooting Table

Table 8–1 suggests corrective actions for certain system problems. When you have a problem, perform the following steps:

- 1. Note the symptoms of the problem.
- 2. Check the **Symptom** column in Table 8–1 for a match.
- 3. Check the conditions for that symptom in the **Possible Cause** column. If more than one possible cause is given, check the possible causes and their suggested solutions in the order listed.
- 4. Follow the advice in the Suggested Solution column.
- 5. If the problem continues, refer to Section 8.13.

Symptom	Possible Cause	Suggested Solution	
InfoServer 1000 Problems			
The InfoServer 1000 Power OK indicator is off.	The ac power cord is not connected. The power outlet may not be working.	Check the power cord connections at both ends. Try a lamp that you know works in the power outlet.	
	The dc power cord is not connected (standalone enclosure only).	Check that the dc power cord is firmly inserted into its socket at the rear of the InfoServer 1000.	
	The [10] power switch is not turned on (1) (InfoTower only.)	Check the setting of the $\square O$ power switch.	
		Remove power from the system. Unplug the terminal cable, network cable, and SCSI cables /terminator. Then plug all cables back in and apply power to the system.	
The power-on display does not appear within 1 second.	The terminal cable is not connected.	Check that the terminal cable is plugged in at both ends.	
		(continued on next page)	

Table 8–1 Troubleshooting Basic System Problems

Symptom	Possible Cause	Suggested Solution
InfoServer 1000 Problems		
	Terminal settings may be incorrect.	Refer to the terminal documentation. The terminal should be set to 9600 baud, 8 bits, and no parity.
The power-on display contains question marks.	Possible CD–ROM drive error.	If question marks appear with any numbers in the power-on display, refer to Section 8.6 for further test instructions.
The server system's software Enter Password: prompt does not appear on the screen.	System software is not installed or is corrupt.	Refer to Section 8.12.
Standalone Expansion Perip	heral Problems	
The power indicator is off.	The 10 power switch is not turned on.	Check the setting of the power switch.
	The power cable or SCSI cable is not installed correctly.	Check that the power cable and SCSI cable (if any) are installed correctly at both ends. Also check the SCSI terminator on the last standalone expansion peripheral.
		InfoTower configuration— Check that the SCSI peripheral is firmly

Table 8–1 (Cont.) Troubleshooting Basic System Problems

Standalone Expansion Peripheral Problems				
The power indicator is off.	The 10 power switch is not turned on.	Check the setting of the power switch.		
	The power cable or SCSI cable is not installed correctly.	Check that the power cable and SCSI cable (if any) are installed correctly at both ends. Also check the SCSI terminator on the last standalone expansion peripheral.		
		InfoTower configuration— Check that the SCSI peripheral is firmly seated in the enclosure (Figure 3–2).		
		Refer to the documenta- tion supplied with your standalone peripheral.		

8.4 Entering the Console Mode

You must be at the **console mode** prompt in order to enter commands or run diagnostics. To enter console mode initially on the InfoServer 1000, you must simulate a failure. The easiest way to do this is to interrupt the network connection during power on as follows:

- 1. Turn the InfoServer 1000 power off.
- 2. Disconnect the network at the rear of the enclosure.
- 3. Turn the InfoServer 1000 power on.

When the system runs its power-on self-test diagnostics, the network test fails and the system will go into console mode (>>>).

- 4. Reconnect the network.
- 5. Enter the desired commands.

Thereafter, you may enable halt-on-break mode with the following command:

>>> SET XDELTA 1

When XDELTA is enabled, you may enter console mode at any time by pressing the Break key on the console. Digital recommends that for normal use, this mode be turned off with the command:

>>>SET XDELTA 0

to avoid halting the InfoServer 1000 with an unintentional console break.

8.5 Interpreting the Power-On Error Messages

The InfoServer 1000 displays important information during its power-on self-test sequence. A single question mark indicates status information. A double question mark indicates possible failures.

___ Note __

If the console terminal does not display any information on the screen after the power-on self-test sequence starts (approximately 1 second), a terminal or system failure may exist. Check Table 8–1, the basic troubleshooting table.

Status Information

A single question mark indicates status information. For example:

If you see a single question mark in the power-on display, but operation of the system seems normal, you can continue to use the system if the question mark is next to a 7 (the SCSI bus controller test).

If you continue to see errors of this type during the power-on display, use the information in the rest of this chapter to solve the problems. If the system continues to display error information or fails to operate normally, call your Digital Services representative.

Possible Failures

If double question marks appear in the power-on display, failures may exist in the system.

A double question mark indicates a hard error that can affect normal operation and use of some system component. For example:
This message indicates that the Ethernet adapter is present in the system, but that one or more of the following problems exist:

- No cable is installed on the Ethernet connector.
- The loopback plug is not installed on the standard Ethernet port for testing purposes.
- The terminator is not installed on the ThinWire Ethernet port.
- A network cable is faulty.

8.6 Running the Self-Tests

Self-test diagnostic programs reside permanently in your system's read-only memory (ROM). You can run self-tests from the console terminal when the system is in console mode (refer to Section 8.4). After running each self-test, enter the command SHOW CONFIG and press Return to display the status of the device tested (refer to Section 8.7).

To test a device in your system, follow these steps:

- 1. Use Table 8-2 to find the number or mnemonic code that represents the device that you want to test.
- 2. Put the system in console mode (refer to Section 8.4).
- 3. Enter TEST (or the letter T) and the mnemonic code or number of the test:

```
>>> T NI (Return)
| |
###
OK
>>>
If you want to !
```

If you want to test a consecutive series of devices, enter TEST (or T) followed by each number in the series separated by a space:

If the self-test is successful, no question mark or error message is displayed, and the console prompt appears.

If a device fails the self-test, a failure message is displayed before the console prompt appears.

>>> T NI Return 1 1 ₽₽₽ ?? 001 6 NI 0172 84 FAIL >>>`

If a self-test results in a failure, execute the SHOW ERROR command to display any error codes.

>>> SHOW ERROR Return ?? 001 6 NI 0172

Mnemonic Code (Number)	Tested Device		
NVR (1)	Nonvolatile RAM		
DZ (2)	Serial line controller (console/LAT port)		
MEM (3)	Memory		
IT (4)	Interval timer		
SYS (5)	System device verification		
NI (6)	Ethernet network interconnect		
SCSI (7)	Internal SCSI bus controller		

Table 8–2 Power-On Display and Self-Test Identifiers

8.7 Displaying the System Configuration Information

The InfoServer 1000 can display configuration, Ethernet ID, and status information. Error information displayed is valuable to your service representative. Have this information ready if you call for service. Refer to Chapter 6 and Appendix C for more information. Table 8-2 lists the mnemonic code used for each device. To display your system configuration, perform the following steps:

- 1. Put the system in console mode (refer to Section 8.4).
- 2. Enter the SHOW CONFIG command at the console prompt (>>>). The following example shows a typical display.

```
>>> SHOW CONFIG [Return]
InfoServer 1000 V2.3-387-V3.7 4-NOV-1993 15:01:21.00
08-00-2B-20-CD-02 ThinWire
4MB
DEVNBR
          DEVNAM
                           INFO
 -----
          ------
                                   -------------
               NVR
     1
                      OK
     2
                DZ
                      OK
     3
               MEM
                      OK
                       4MB = SO = 4MB, S1 = 0MB
     4
                IT
                      OK
     5
               SYS
                      OK
     6
                      OK
                NI
     7
              SCSI
                      OK
                      1-RRD42
                                2-RZ23L
                                           6-INITR
```

Solving Problems

8.8 Displaying the Device Information

To determine which devices are installed in your system and their SCSI bus ID assignments, enter the SHOW DEVICE command at the console prompt (>>>). The system displays information similar to the following:



8.9 Running the Test Utilities for Expansion Peripherals

If an expansion peripheral fails, check that all cable connections are in place.

To test a CD-ROM drive, follow these steps:

- 1. Put the system in console mode (refer to Section 8.4). The console prompt (>>>) appears on your screen.
- 2. Insert a disc in the CD-ROM drive, following the instructions in Appendix B.
- 3. Enter the TEST SCSI command at the console prompt.
- 4. The test should display the normal response code as follows:

```
>>> T SCSI Return

| |

###

OK

>>>

If the SCSI test fails, issue the following commands:
```

>>> U >>> SHOW DEVICE

Check the display to ensure that all devices are present. If connected devices do not display on the screen, check that the SCSI IDs are unique for each device and that each device is powered up. If the peripheral device fails to display after you have checked these possible problems, call your Digital Services representative for help.

8.10 Restarting the System After Running Tests

While running any of the tests or procedures in this chapter, your system is in console mode. If the InfoServer 1000 system software is already loaded in the internal memory (refer to Section 5.2), resuming normal operation is easy. Reenter **program mode** in one of two ways:

- Enter the BOOT command at the console prompt and press Return. The system then searches its internal memory for the server's load image.
- Turn the InfoServer 1000 off and then on again. The server is configured to boot automatically at power-on.

Either method lets the server boot the system software in the internal memory. If the software has not been copied into memory, place the InfoServer software compact disc in the compact disc drive. If the system fails to boot initially from the internal memory, the system automatically tries to boot from the compact disc drive. It will try to boot indefinitely.

If the system does not boot when you enter the BOOT command, refer to Section 8.12.

8.11 Reestablishing the Console Parameters

It may be necessary to reestablish key console parameters if you have replaced the system. Enter the following commands at the console prompt:

>>> SET BOOT DKA200, DKA100, FLASH Return >>> SET HALT 2 Return >>> SET BFLG D0000000 Return >>> SET XDELTA 0 Return >>> BOOT Return

8.12 Recovering from Bootstrap Failures

If the InfoServer 1000 system fails to boot and displays the following error message, the internal FLASH memory is corrupted:

```
- IMAGE NOT PRESENT C7
```

Use the following procedure to upgrade FLASH memory. Otherwise, the procedure should be used *only* if the InfoServer 1000 software in FLASH is known to be out-of-date as described in the software release notes.

- 1. Turn the InfoServer 1000 power off.
- 2. Disconnect the Ethernet cable at the rear of the enclosure.
- 3. Connect the console terminal, turn it on, and allow it to warm up.
- 4. Turn the InfoServer 1000 power on.
- 5. If the self-test diagnostics do not complete and the console prompt >>> does not appear, proceed to step 11.
- 6. If you have a CD-ROM drive on the system, execute a SHOW DEVICE command and make note of the unit number of the CD-ROM drive. Otherwise, proceed to step 10.
- Boot the InfoServer 1000 from the CD-ROM drive with the command:
 >>> B DKAnnn:

8-12

where nnn refers to the unit number.

- 8. When the Enter Password: prompt appears, enter the default password (ESS) or the last saved password. If you have forgotten the current password, refer to the troubleshooting chapter in the *InfoServer System* Operations Guide.
- 9. Enter the following commands in response to the InfoServer> prompt:

InfoServer> SET STATE OFF InfoServer> UPDATE SYSTEM DKm: FLASE

where m is the unit number of the device containing the compact disk.

Proceed to step 13.

10. If there is no CD-ROM drive available, enter the following command to boot the InfoServer 1000 from the Ethernet:

>>> B/100 ESA0

Bootfile:

Proceed to step 12 if the Bootfile: prompt appears.

- 11. If the B/100 ESA0 command does not produce a Bootfile prompt, or if the InfoServer 1000 does not even execute the power-on self-test to the point where the >>> prompt appears, then perform the following steps:
 - a. Turn the power off.
 - b. Remove two retaining screws on the bottom and slide off the InfoServer 1000 cover.
 - c. Move the external jumper (J1) to the "enabled" position (Figure 8-1).
 - d. Replace the cover and turn the power on. The Bootfile: prompt should appear.
- 12. At the Bootfile: prompt, enter the name of the boot image provided by the software release notes document. If step 11 has been executed, then the boot image should be chosen to update the diagnostics as well as the InfoServer 1000 firmware image.

You must have the image available for MOP downline loading from another InfoServer 1000 or from a host system. Refer to the *InfoServer System Operations Guide* for instructions on enabling MOP downline loading of InfoServer update images.

Solving Problems





13. The following display appears:

---InfoServer 1000 FLASH ROM Update Program V1.5---Current Contents of FLASE: Segment Size Version Maj.Rev. Min.Rev. Pat.Rev. Time Name 0000001 00004000 0000001 00000002 00010000 00000001 00000003 00005000 0000002 0000001 00000071 00000000 0000000 EXT HEADER 00000004 0003c000 00000002 0000000F 00000007 00000000 2B4D6498 SYSLOCS 00000005 0000F000 00000002 0000000F 00000007 00000000 2B4D6498 HELP.MEM Update Contents of FLASH: Version Maj.Rev. Min.Rev. Pat.Rev. Time Segment Size Name 00000001 00004000 00000001 00000002 00010000 00000001 00000003 00005000 0000002 0000001 00000071 00000000 00000000 EXT HEADER 00000004 0003C000 00000002 000000F 00000007 00000000 2B4D6498 SYSLOCS 00000005 0000F000 0000002 000000F 0000007 0000000 2B4D6498 HELP.MEM --- Flash image size (HEX): 00050000 Starting offset (HEX): 00020000 ---CAUTION------- EXECUTING THIS PROGRAM WILL CHANGE YOUR CURRENT INFOSERVER ROM IMAGE ---Do you really want to continue [Y/N] ? :

If the SYSLOCS and HELP. MEM segments do not appear in the "Current Contents" display, then FLASH memory is indeed corrupt and you may continue with the process. If the two unlabeled segments do not appear, then the diagnostic code is corrupt. The diagnostic update displays the following information. It is important to note that the starting offset is zero.

--- Flash image size (HEX): 00070000 Starting offset (HEX): 00000000 ----CAUTION------ EXECUTING THIS PROGRAM WILL CHANGE ALL OF YOUR CURRENT ROM ----Do you really want to continue [Y/N] ? :

8-15

Solving Problems

14. Respond to the question with a Y or an N. If N is chosen, proceed to step 15. Otherwise, the following display appears.

```
Do you really want to continue [Y/N] ? : Y
DO NOT ATTEMPT TO INTERRUPT PROGRAM EXECUTION!
DOING SO WILL RESULT IN LOSS OF OPERABLE STATE!
 The program will take one minute at most.
 led codes display information
0000 0nnn
                       in process of erasing
                                                FEPROM block nnn
                   -
0000 1nnn
                       in process of programming FEPROM block nnn
                       in process of verifying FEPROM block nnn
0001 0nnn
                   -
DO NOT POWER DOWN OR TOUCH BREAK, OR YOU WILL DAMAGE THE CPU MODULE !!!!
 !!!! THE SYSTEM WILL THEN DO A POWERUP RESTART and HALT at the console!!!!
 !!!! OTHERWISE, POWER CYCLE THE BOX AFTER ALL LEDS ARE LIT !!!!
Erasing
            Block 0000005
Programming Block 0000005
Verifying Block 00000005
           Block 0000006
Erasing
Programming Block 0000006
Verifying Block 0000006
           Block 0000007
Erasing
Programming Block 0000007
Verifying Block 0000007
?06 HLT INST
 PC= 00000B7A PSL= 041F0004
```

>>>

In some cases, the InfoServer 1000 may hang after the verification of block 7 with all of the indicators lit. If this happens, turn the InfoServer 1000 power off and then on.

- 15. If step 11 has been executed, turn the InfoServer 1000 power off, remove the cover, restore the external jumper (J1) to its original position, and replace the cover.
- 16. Restore the halt action with the following command and boot the system:

>>> SET HALT 2 >>> B/O FLASH

17. Refer to the *InfoServer System Operations Guide* to restore functions if the licensed functions do not appear.

1-800-354-9000

8.13 Service Information

If you have followed the corrective actions listed in this chapter and continue to have problems with your system, call your Digital Services representative.

Before you call:

1. Write down the serial and model numbers of your system. Your system is identified by a label on the rear of the InfoServer 1000 or InfoTower. For example:

Model: SEAxx-Ax, InfoServer 1000 system.

Your Digital Services representative may need this number when you call.

- 2. Make notes based on the troubleshooting table at the beginning of this chapter. This information informs your service representative as to the state of your system when the problem occurred.
- 3. Be prepared to read information from the screen and to enter commands at the keyboard while you talk to your service representative on the telephone.
- 4. Six indicators on the front of the InfoServer 1000 show key status information. Write down the status of these indicators. Your service representative may ask you to describe which indicators are on or off.

8.14 Command Summary

Table 8-3 and Table 8-4 list the SET and SHOW command options used in these troubleshooting procedures. To display a list of all InfoServer commands, enter HELP at the console prompt and press Return. For more information on commands, refer to the InfoServer System Operations Guide.

Command	Default Set	
SET BFLG	Sets the bootstrap address.	
SET BOOT	Sets the bootstrap device.	
SET HALT	Sets the default halt action.	
SET SCSI	Sets the SCSI device ID number.	
SET XDELTA	Sets the enable/disable halt on Break key.	

Table 8–3 SET Command Summary

Solving Problems

Command	Information Displayed
SHOW BFLG	The bootstrap address
SHOW BOOT	The bootstrap device
SHOW CONFIG	Internal device configuration
SHOW DEVICE	Devices available and SCSI ID
SHOW ETHER	Hardware Ethernet address
SHOW HALT	Default action after your system halts
SHOW MEMORY	Memory for your InfoServer 1000
SHOW SCSI	SCSI device ID number
SHOW XDELTA	Halt on Break key status

 Table 8-4
 SHOW Command Summary

. . . .

A

Installation Checklist

A.1 Introduction

This appendix provides a checklist of the basic installation steps. For more detailed instructions refer to the following chapters:

- Chapter 2 for standalone configurations such as the InfoServer VXT or InfoServer Scribe
- Chapter 3 for InfoServer Librarian configurations using the InfoTower enclosure
- Chapter 4 for the InfoServer SW1000 installed in a StorageWorks pedestal
- Chapter 5 concludes the installation

The applicable section number is identified in parenthesis.

A.2 Standalone InfoServer 1000 Installation Steps

Follow these basic steps to install standalone InfoServer 1000 system configurations.

- 1. Choose a suitable location for the system (Section 2.2).
- 2. Unpack and inspect the standalone system components (Section 2.3).
- 3. Identify the InfoServer 1000 ports and connectors (Section 2.3.2).
- 4. Attach the Ethernet terminator or loopback plug (Section 2.3.3).
 - Connect the ThinWire Ethernet terminator (Section 2.3.3.1), or
 - Connect the standard Ethernet loopback plug (Section 2.3.3.2).
- 5. Connect a console terminal (Section 2.3.4).
- 6. Connect the standalone expansion peripherals (Section 2.3.5).
- 7. Connect the InfoServer 1000 power supply and power cord (Section 2.3.6).
- 8. Familiarize yourself with the front panel indicators (Section 1.5).

Installation Checklist

9. Turn on the system (Section 2.4 and Section 5.2).

InfoServer 1000 software is factory-loaded. The software compact discs shipped with the system are for backup purposes. Do not attempt to load software as part of the installation process.

CAUTION

10. Load a compact disc or tape, if applicable (Appendix B).

11. Connect the system to a network (Section 5.3).

- 12. Boot the system (Section 5.4).
 - Boot automatically without a console (Section 5.4.1) or manually with a console (Section 5.4.2).

A.3 InfoTower Installation Steps

Follow these basic steps to install the InfoServer Librarian system in an InfoTower enclosure.

1. Choose a suitable location for the system (Section 3.2).

_____ CAUTION _____

Because of the weight of the InfoTower system (25.5 kg (56 lb) fully loaded), two people should lift the equipment out of the shipping carton and place it on a work surface.

2. Unpack and inspect the InfoTower system components (Section 3.3).

_____ WARNING _____

For the vertical installation of the InfoTower, use an enclosure mounting stand.

_____ Note _____

After moving the system into position, unlock and open the enclosure door. Ensure that each peripheral is seated properly and is firmly connected to the backplane by pressing firmly on the front bezel.

- 3. Identify the InfoTower ports and connectors (Section 3.3.2).
- 4. Install the SCSI terminator (Section 3.3.3).
- 5. Attach the Ethernet terminator or loopback plug (Section 3.3.4).
 - Connect the ThinWire Ethernet terminator (Section 3.3.4.1), or
 - Connect the standard Ethernet loopback plug (Section 3.3.4.2).
- 6. Connect a console terminal (Section 3.3.5).
- 7. Connect the InfoTower system power cord (Section 3.3.6).
- 8. Familiarize yourself with the front panel indicators (Section 1.5).
- 9. Turn on the system (Section 3.4 and Section 5.2).

_ CAUTION _____

InfoServer 1000 software is factory-loaded. The software compact discs shipped with the system are for backup purposes. Do not attempt to load software as part of the installation process.

- 10. Load a compact disc (Appendix B).
- 11. Connect the system to a network (Section 5.3).
- 12. Boot the system (Section 5.4).
 - Boot automatically without a console (Section 5.4.1) or manually with a console (Section 5.4.2).

A.4 InfoServer SW1000 Installation Steps

Follow these basic steps to install the InfoServer SW1000 system in a StorageWorks enclosure. To install the pre-configured StorageWorks InfoServer SW1000, go to step 6.

- 1. Unpack and inspect the InfoServer SW1000 system components (Section 4.3).
- 2. Identify the InfoServer SW1000 ports and connectors (Section 4.4.1).
- 3. Determine the SCSI bus termination and configuration (Section 4.4.2).
- 4. Insert the InfoServer SW1000 into the StorageWorks backplane (Section 4.4.3).

Installation Checklist

- 5. Connect a console terminal (Section 4.4.5).
- 6. Attach the ThinWire Ethernet terminator (Section 4.4.6).
- 7. Turn on the system (Section 4.5 and Section 5.2).

_____ CAUTION _____

InfoServer SW1000 software is factory-loaded. The software compact discs shipped with the system are for backup purposes. Do not attempt to load software as part of the installation process.

- 8. Load a compact disc (Appendix B).
- 9. Connect the system to a network (Section 5.3).
- 10. Boot the system (Section 5.4).

Boot automatically without a console (Section 5.4.1) or manually with a console (Section 5.4.2).

Loading and Unloading Compact Disks

B.1 Introduction

The information in this appendix provides instruction on how to load and unload CD-ROM disks into the reader using a caddy.

B.2 Loading a Compact Disc

When you load a disc into the compact-disc drive, insert the entire disc **caddy** into the disc door on the drive. Do not remove the disc from the caddy.

To load compact discs that you want to serve to the network, follow these steps:

- 1. Examine the disc caddy. Ensure that it is not cracked or damaged. Never load a damaged caddy into a compact-disc drive.
- 2. If a protective film is on the lid of the caddy, remove the film before using the caddy (Figure B-1).
- 3. To open the lid of the caddy, press the tabs on both sides and lift the lid (Figure B-1).
- 4. To load an empty caddy with a disc, hold the disc by the edges and place the disc (with disc label up) into the caddy (Figure B-2). Do not touch the surface of the disc. Be sure to place the disc beneath the edge of the caddy.
- 5. Press both corners firmly to close the lid (Figure B-2).

. Note ..

The loading procedure for the standalone and enclosure-mounted drives is the same with one exception; the standalone drive requires setting the power switch to on. Figure B-3 shows the standalone drive.

6. Insert the caddy into the drive (Figure B-3).

The green activity light should turn on within five (5) seconds. If the drive accepts the disc and caddy, but the light does not come on, the disc could be placed incorrectly in the caddy. Remove the disc and check that the label is facing up when you insert the disc into the drive again. If the activity light still does not come on, there could be a problem with the drive. Contact your local Digital Services representative.

The drive reads the Table of Contents (TOC) on the disc. The busy indicator lights while the TOC is read. When the busy indicator light goes off, the drive is ready to receive commands.

Figure B-1 Opening the Caddy



Figure B-2 Loading the Disc





Loading and Unloading Compact Disks





SHR-XR0074-90A

B.3 Unloading a Compact Disc

Before unloading a compact disc, ensure that the green activity light is not flashing. If the light is flashing, the compact-disc drive is transferring data. Wait until the green activity light stops flashing.

Press the eject button on the drive to eject the caddy as shown in Figure B-4.

Figure B-4 Ejecting the Caddy from the Drive



SHR-XR0075-90A

The eject button on the drive does not function when disabled by software or if power to the drive is turned off. Use the following procedure for manual ejection:

- 1. Turn off the power to the drive.
- 2. Insert a steel rod, about 1.2 millimeters in diameter and 35 millimeters or 1.37 inches long, and push it into the emergency eject hole at the lower right of the front panel (Figure B-5). The rod can be made by straightening a large paper clip.

•



.

Figure B-5 Front Panel of the Drive

-

C SCSI ID Numbers

C.1 Introduction

The information in this appendix is for technical users who want to create custom configurations.

The small computer systems interface (SCSI) is an interface designed for connecting disks and other peripheral devices to computer systems. SCSI is defined by an American National Standards Institute (ANSI) standard and used by many computer and peripheral vendors throughout the industry. The InfoServer 1000 complies with the SCSI-II standard.

Up to eight (8) SCSI devices can share a SCSI bus. All data and commands are sent back and forth on the bus (cable). Each SCSI device attached to the cable "looks" at all the data, but a SCSI device recognizes only the data that has the proper device identification, called a SCSI ID number. ¹

C.2 SCSI ID Default Settings

You are responsible for the SCSI ID settings on your equipment. Digital sets each SCSI device to a default setting before the equipment leaves the factory. You may never need to change a default setting. Default settings should only be changed when a system is configured with more than one of a particular device.

The SCSI ID numbers are 0 through 7. The number 7 is reserved for a device requiring the highest priority on the SCSI bus. The number 6 is reserved for the controller. The remaining six ID numbers (0 through 5) are available for custom configuration.

Table C-1 lists the default SCSI settings for the InfoServer 1000 system. The ID numbers listed give optimal performance on most systems. However, if you have special performance needs, you can set the ID number on each SCSI device manually. For most applications, SCSI ID numbers can be set arbitrarily as long as no two devices share the same ID.

¹ The SCSI controller within the InfoServer 1000 occupies one SCSI ID number.

SCSI ID Numbers

SCSI ID	SCSI Device
0	First CD–ROM drive
1	Available for expansion
2	Available for expansion
3	Available for expansion
4	Available for expansion
5	Available for expansion
6	SCSI controller ¹
7	Available for expansion (highest priority) ¹
1	

Table C-1 SCSi ID Numbers for InfoServer 1000 System

¹In the case of InfoServer SW1000, SCSI ID 6 is open, and SCSI ID 7 is the SCSI controller.

_____ Note _____

Each device has a set of switches or jumpers that can be set for a specific SCSI ID. The optional standalone hard-disk drives and additional compact-disc drives are mounted in expansion enclosures that have external switches.

C.3 Determining the SCSI ID Numbers

To determine SCSI ID numbers, enter the SHOW DEVICE command at the console prompt (>>>), then press Return. For directions on how to place the system in console mode, refer to Section 8.4. The command displays your system configuration on the screen.

In the following example, the system contains a controller, an RRD42 compact-disc drive, and an RZ23L hard-disk drive. There are five open ID numbers.

>>> SHOW DEV	LCE Herom						
VMS/VMB	ADDR	DEVTYPE	NUMBYTES	RM/FX	WP	DEVNAM	REV

ESA0	08-00-2	B-20-CD-02 1	hinWire				
DKA100	A/1/0	DISK	121.64MB	FX		RZ23L	2528
DKA200	A/2/0	DISK	593.49MB	RM	WP	RRD42	1.2b
DKA300	A/3/0	DISK	600 MB	RM	WP	RRD43	0064
HostID	A/6	INITR					
0	2	3	4	6	6	7	8

Device name

2 Controller/SCSI ID/logical number

In this example: SCSI ID 0 - Open SCSI ID 1 - RZ23L hard-disk drive SCSI ID 2 - RRD42 CD-ROM drive SCSI ID 3 - RRD43 CD-ROM drive SCSI ID 4 — Open SCSI ID 5 - Open SCSI ID 6 — Internal SCSI controller (open for SW1000) SCSI ID 7 — Open (Internal SCSI controller for SW1000)

6 Generic device type



6 Removable/fixed media

- **6** Write protected
- Media revision level
- Drive firmware revision

Rules for Setting SCSI ID Numbers

- You can have up to seven (7) expansion peripherals on the SCSI bus.
- Each peripheral must have its own unique identifier.
- The SCSI bus must be terminated.

Proper operation of the SCSI bus requires that you use high-quality, properly configured cables and connectors to connect all devices. Digital recommends that you use only Digital-supplied cable assemblies intended for interconnecting SCSI devices. This ensures that the impedance characteristics, signal propagation velocity, inductance, capacitance, cross-talk, grounding, conductor pairing, and shielding meet the requirements for proper operation of the bus.

Digital also recommends that all units on the SCSI bus be powered from a common ac power source. Digital does not guarantee the proper operation of any SCSI bus that uses cable assemblies not supplied by Digital, or that is not configured in accordance with Digital's recommendations.

System Specifications

D.1 Introduction

This appendix lists InfoServer 1000, InfoTower, and InfoServer SW1000 system specifications, including operating and storage conditions.

D.2 InfoServer 1000 Hardware Specifications

Table D-1 lists hardware specifications for the standalone InfoServer 1000.

Specification	Value or Description 32-bit CPU with 1 KB of first-level cache		
Processor			
DRAM memory	4 MB		
ROM memory	512 KB flash, 32 KB NVR		
Optional storage	CD–ROM expansion peripherals and other standard SCSI-II disk and tape devices		
Terminals	Supports the VT300 and VT400 series		
Interfaces	1 SCSI port 1 asynchronous 9600 baud terminal port 1 Ethernet port (ThinWire or standard (thickwire))		

Table D-1 InfoServer 1000 System Hardware Specifications

D.3 Standalone InfoServer 1000 Specifications

Table D-2 lists the standalone InfoServer 1000 power requirements and specifications.

Specification	Value or Description
H7827-BA input voltage	Automatically adjusting AC input. Range: 100 Vac to 120 Vac, 220 Vac to 240 Vac
H7827-BA input current (typical)	0.7 A in a 110 Vac circuit 0.42 A in a 220 Vac circuit
Power (maximum)	35 W
Frequency	47 Hz to 63 Hz
H7827-BA output voltage/current	5.1 Vdc @ 2.5 A 12.1 Vdc @ 0.875 A
H7827-BA DC output cable length	1.83 m (6.0 ft)
H7827-BA DC output cable connector	7-position mini-DIN 5 V pins 1, 3 12 V pins 2, 4 return pins 5, 6, 7

Table D-2 Standalone InfoServer 1000 Power Specifications

Table D-3 lists the standalone InfoServer 1000 weight and dimensions.

Device	Weight	Height	Width	Depth	
InfoServer 1000	1.25 kg	3.90 cm	14.5 cm	21.1 cm	
	2.75 lb	1.53 in	5.70 in	8.30 in	
H7827-BA power supply	0.8 kg	5.16 cm	9.00 cm	15.75 cm	
	1.75 lb	2.03 in	3.54 in	6.20 in	

Table D--3 Standalone InfoServer 1000 Weight and Dimensions

Table D-4 lists the standalone InfoServer 1000 operating conditions.

Table D-4 Standalone InfoServer 1000 Operating Conditions

Specification	Value
Temperature range	10° C to 40° C (50° F to 104° F)
Temperature change rate	11° C/hour (20° F/hour) maximum
Relative humidity	10% to 80% noncondensing
Maximum wet bulb temperature	28° C (82° F)
Minimum dew point	2° C (36° F)
Altitude	2400 m at 36° C (8000 ft at 96° F)

Table D-5 lists the standalone InfoServer 1000 storage conditions.

 Table D-5
 Standalone InfoServer 1000 Storage Conditions

Specification	Value		
Temperature range	5° C to 50° C (41° F to 122° F)		
Relative humidity	10% to 95% at 66° C (151° F) (noncondensing)		
Altitude	0 to 2400 m (0 to 8000 ft)		
Maximum wet bulb temperature	32° C (90° F)		
Minimum dew point	2° C (36° F)		

System Specifications

Table D–6 and Table D–7 list the acoustic specifications for the InfoServer 1000 system.

	Sound Pow	er Levei	Sound Pres	ssure Level
	L _{WAd} , B		L _{pAm} , dBA	
			Bystander	Positions
	ldle	Operate	ldle	Operate
SEADB-AA		No acc	oustic noise	
H7827-BA		No acc	oustic noise	
RRD42-FA	3.9	4.0	21	22
RRD43	<3.8	<3.8	<20	<20
RRD44	<3.8	<3.8	<20	<20
SZ03B-BC	5.2	5.2	38	38

ډ,

Table D-6 InfoServer 1000 System Acoustic Values (Declared Values Per ISO 9296 and ISO 7779)

Note: Current values for specific configurations are available from Digital Services representatives. 1 B = 10 dBA.

	Schalleistungspegel		Schalldruck	pegei		
	L _{WAd} , B	L _{wad} , B		L _{pAm} , dBA		
				ositionen		
	Leerlauf	Betrieb	Leerlauf	Betrieb		
SEADB-AA	keine meßbaren Schallemissionen					
H7827-BA	keine meßbaren Schallemissionen					
RRD42-FA	3,9	4,0	21	22		
RRD43	<3,8	<3,8	<20	<20		
RRD44	<3,8	<3,8	<20	<20		
SZ03B-BC	5,2 5,2 38 38					

Table D-7 InfoServer 1000 Schallemissionswerte (Werteangaben nach ISO 9296 und ISO 7779/DIN EN27779)

Aktuelle Werte für spezielle Ausrüstungsstufen sind über die Digital Equipment Vertretungen erhältlich. 1 B = 10 dBA.

D.4 InfoTower System Specifications

Table D-8 lists the functional specifications for the InfoTower system.

Specification	Value or Description				
System enclosure	Desktop and Tower configuration				
Storage capability	Seven 5.25-inch (13.3 cm) half-height SCSI devices				
Interface	SCSI-II				
Nominal AC voltage	120/240 Vac switch selectable				
Power source phasing	Single phase				
Nominal frequency	50 Hz—60 Hz				
Voltage ranges	90 Vac—135 Vac 180 Vac—265 Vac				
Line frequency tolerance	47 Hz-63 Hz				
Maximum AC running current	5 A @ 115 Vac 3 A @ 230 Vac				
Maximum power consumption	307.0 W				
Height	17.78 cm (7.0 in.)				
Width	43.18 cm (17.0 in.)				
Depth	43.18 cm (17.0 in.)				
Weight	23.64 kg (52 lb) loaded with 7 CD–ROM drives				
Noncondensing maximum operating altitude	2400 m (8000 ft)				

Table D-8 InfoTower System Functional Specifications

Table D-9 lists the physical specifications for the InfoTower system.

Specification	Operating	Nonoperating		
Ambient temperature	10° C to 35° C	-40° C to 66° C		
Relative humidity	10% to 85%	10% to 95% (packaged) noncondensing		
Maximum wet bulb	28° C (82° F) (noncon- densing)	32° C (90° F) (packaged) (noncondensing)		
Altitude	2400 m (8000 ft)	5000 m (16000 ft)		
Heat dissipation	1,050 Btu/h	N/A		
Temperature gradient	11° C/hr (20° F/hr)	20° C/hr (36° F/hr)		

 Table D-9
 InfoTower System Physical Specifications

Table D-10 lists the input electrical specifications for the InfoTower system.

Specification	Value or Description			
Line voltage	90 Vac—135 Vac, 180 Vac—265 Vac (switch selectable)			
Line frequency	47 Hz—63 Hz			
Line current	8 A maximum @ 90 Vac including 3 A auxilliary 6 A maximum @ 180 Vac including 2 A auxilliary			
Power consumption	60 W @ minimum load			
Apparent power (maximum)	462 VA @ maximum load, 120 Vac range 507 VA @ maximum load, 220 Vac range			
Power factor (typical)	0.5			
Inrush current (typical)	13.6 A @ 120 Vac, 9.9 A @ 220 Vac			
Efficiency (typical)	65%			

Table D-10 InfoTower System Input Characteristics—Electrical Specifications

System Specifications

Table D-11 lists the output electrical specifications for the InfoTower system.

		-			
Nominal	Load Current	Regulation Tolerance			
Output Voltage	Maximum/Minimum	Maximum/Minimum			
5 Vdc	20 A/3 A	5.25 V / 4.80 V			
12 Vdc	8 A / 0 A	12.60 V / 11.40 V			
-5 Vdc	0.5 A / 0 A	-5.50 V / -4.50 V			
-12 Vdc	0.5 A / 0 A	-13.20 V / -10.80 V			

Table D-11 InfoTower System Output Characteristics-Electrical Specifications

Table D-12 and Table D-13 list the acoustic specifications for the InfoTower system.

	Sound Power	Level	Sound Pressure Level			
	L _{WAd} , B		L _{pAm} , dBA			
			Bystander Positions			
	ldle	Operate	idie	Operate		
SEADB-PA (0 x CD-ROM)	5.2	5.2	36	36		
SEADC-PA (0 x CD-ROM)	5.2	5.2	36	36		
SEADB-RA/RE (4 x CD–ROM)	5.2	5.2	36	36		
SEADC-RA/RE (4 x CD-ROM)	5.2	5.2	36	36		
SEADB-SA/SE (7 x CD-ROM)	5.2	5.2	36	36		
SEADC-SA/SE (7 x CD-ROM)	5.2	5.2	36	36		

Table D-12	InfoTower System	Acoustic	Values	(Declared	Values	Per	ISO	9296
	and ISO 7779)							

Note: Current values for specific configurations are available from Digital Services representatives. 1 B = 10 dBA.
	Schalleistungspegel		Schalldruckpegel	
	L _{WAd} , B		L _{pAm} , dBA	
			Zuschauerpositionen	
	Leerlauf	Betrieb	Leerlauf	Betrieb
SEADB-PA (0 x CD-ROM)	5,2	5,2	36	36
SEADC-PA (0 x CD-ROM)	5,2	5,2	36	36
SEADB-RA/RE (4 x CD–ROM)	5,2	5,2	36	36
SEADC-RA/RE (4 x CD-ROM)	5,2	5,2	36	36
SEADB-SA/SE (7 x CD-ROM)	5,2	5,2	36	36 .
SEADC-SA/SE (7 x CD–ROM)	5,2	5,2	36	36

Table D-13	InfoTower Schallemissionswerte (Werteangaben nach ISO 9296
	und ISO 7779/DIN EN27779)

Aktuelle Werte für spezielle Ausrüstungsstufen sind über die Digital Equipment Vertretungen erhältlich. 1 B = 10 dBA.

D.5 InfoServer SW1000 Specifications

Table D-14 lists the weight and dimensions for the InfoServer SW1000 system.

Device	Weight kg (lbs)	Height mm (in)	Width mm (in)	Depth mm (in)	
5.25-inch SBB	1.93 (4.25)	121 (4.8)	152 (6.0)	267 (10.5)	

Table D-14 InfoServer SW1000 Weight and Dimensions

Table D-15 lists the InfoServer SW1000 power requirements and specifications.

Table D–15 InfoServer SW1000 Power Spe	Decifications
--	----------------------

Specification	Value or Description	
SWDCI-AA Input voltages	12 Vdc @ .7A 5 Vdc @ 2.5A	
	5 Vdc @ 2.5A	

Table D-16 and Table D-17 list the acoustic specifications for the InfoServer SW1000 system.

Table D-16 InfoServer SW1000 System Acoustic Values (Declared Values Per ISO 9296 and ISO 7779)

	Sound Power Level L _{WAd} , B		Sound Pressure Level L _{PAm} , dBA Bystander Positions	
	ldle	Operate	ldie	Operate
SWDCI-AA	No acous		ustic noise	

Note: Current values for specific configurations are available from Digital Services representatives. 1 B = 10 dBA.

Table D-17 InfoServer SW1000 Schallemissionswerte (Werteangaben nach ISO 9296 und ISO 7779/DIN EN27779)

	Schalleistungspegel		Schalldruckp	egel
	L _{WAd} , B		L _{pAm} , dBA	
			Zuschauerpositionen	
	Leerlauf	Betrieb	Leerlauf	Betrieb
SWDCI-AA		keine meßbare	n Schallemissionen	
SWDCI-AA	Keine mebbaren		n Schallemissionen	

Aktuelle Werte für spezielle Ausrüstungsstufen sind über die Digital Equipment Vertretungen erhältlich. 1 B = 10 dBA.

Table D-18 lists the InfoServer SW1000 operating conditions.

Table D-18 Storage Works InfoServer SW1000 Operating Conditions

Specification	Value
Temperature range	18° C to 24° C (65° F to 75° F)
Temperature change rate	3° C/hour (5.4° F)
Temperature step rate	3° C/hour (5.4° F)
Relative humidity	40% to 60% noncondensing with a step change of 10% or less (noncondensing)
Altitude	From sea level to 2400 m (8000 ft)
Air quality	Maximum particle count .5 micron or larger, not to exceed 500,000 particles per cubic ft of air
Inlet air volume	.026 cubic m per second (50 cubic ft per minute)
Maximum wet bulb temperature	28° C (82° F)
Minimum dew point	2° C (36° F)

Table D-19 lists the StorageWorks InfoServer SW1000 storage or nonoperating conditions.

Table D-19	StorageWorks InfoServer SW1000 Storage and Nonoperati	ng
	Conditions	

Specification	Value
Temperature range	-40° C to 66° C (40° F to 151° F)
Relative humidity	8% to 95% in original shipping container (noncondensing); otherwise, 50% (noncondensing) (151° F) (noncondensing)
Altitude	-300m (-1000 ft) to 3600 m (+12,000 ft) MSL

System Specifications

Table D-20 lists the StorageWorks InfoServer SW1000 thermal stabilization specifications.

Ambient Temperature Range °C	Ambient Temperature Range °F	Minimum Stabilization Time	
60 to 66	140 to 151	3 hours	
50 to 59	122 to 139	2 hours	
40 to 49	104 to 121	1 hours	
30 to 39	86 to 103	30 minutes	
18 to 29	65 to 85	None	
10 to 17	50 to 64	30 minutes	
0 to 9	32 to 49	1 hour	
-10 to -1	14 to 31	2 hours	
-20 to -11	-4 to 13	3 hours	
-30 to -21	-22 to -5	4 hours	
-40 to -31	-40 to -21	5 hours	

 Table D-20
 StorageWorks InfoServer SW1000 Thermal Stabilization

 Specifications
 Specifications

Ε

Field-Replaceable Units

E.1 Introduction

This appendix contains a list of all the field-replaceable units (FRUs) and their part numbers. Table E-1 lists FRUs for the InfoServer 1000 system. Table E-2 lists FRUs for the InfoTower system. Table E-3 lists the ac power cables.

Part Number	FRU Description
12-22196-01	Plug, thickwire loopback
12-25869-01	T-connector, coaxial, BNC (H8223)
12-26318-01	Terminator, 50-ohm coaxial BNC (H8225)
12-35759-01	Terminator, SCSI-II
17-00811-03	Console cable (BC16E-25)
17-01351-01	Cable assembly, SCSI 50-pin, 3 ft (BC19J-03)
17-01351-04	Cable assembly, SCSI 50-pin, 1.5 ft (BC19J-1E)
17-02641-01	Cable assembly, SCSI-II to Champ, 3 ft (BC09D-03)
17-02641-06	Cable assembly, SCSI-II to Champ, 1 ft (BC09D-01)
17-03579-02	Cable assembly, ribbon, 16 inch, 34-conductor
17-03712-01	Cable assembly, 4-conductor, drive bracket dc power
17-03713-01	Cable assembly, 4-conductor, 18 AWG dc power extender
17-03714-01	Cable assembly, 50-conductor, drive bracket SCSI
30-34512-01	CD–ROM drive caddy
54-21868-01	Network adapter, thickwire
54-21870-01	Network adapter, ThinWire
70-30343-01	Standalone InfoServer 1000, ThinWire
70-30343-03	Standalone InfoServer 1000, thickwire
70-30343-04	InfoServer 1000 configured for InfoTower (Digital)
70-31732-01	InfoServer SW1000 sub-assembly
74-47025-01	Bezel extraction tool
BA35x-MB	SCSI terminator board
BA35x-MC	SCSI jumper board
H7827-BA	Power supply for standalone InfoServer 1000

.

Table E-1 InfoServer 1000 Field-Replaceable Units

Part Number	FRU Description		
12-30552-01	Terminator, SCSI		
17-02008-01	Cable assembly, SCSI 68-pin to 50-pin, 3 ft (BC56H-03)		
30-40426-01	Stand		
70-30871-01	Drive mounting bracket assembly, 1/2-height		
SZ18Z-AF	InfoTower option (empty)		

Table E-2 InfoTower Field-Replaceable Units

Table E-3 AC Power Cables

Part Number	FRU Description			
17-00198-14	Power cable, ac, Australia/New Zealand (BN19H-2E)			
17-00199-21	Power cable, ac, Central Europe (BN19W-2E) ¹			
17-00209-15	Power cable, ac, UK/Ireland (BN19A-2E)			
17-00210-13	Power cable, ac, Switzerland (BN19E-2E)			
17-00310-08	Power cable, ac, Denmark (BN19K-2E)			
17-00364-18	Power cable, ac, Italy (BN19M-2E)			
17-00456-16	Power cable, ac, India/S. Africa (BN19S-2E)			
17-00457-16	Power cable, ac, Israel (BN18L-2E)			
17-00606-02	Power cable, ac, US/Canada/Japan (BN19P-1K)			

¹Austria, Belgium, Finland, France, Germany, Netherlands, Norway, Sweden, Hungary, Poland, Czechoslovakia, Portugal, and Spain

-

Glossary

32-bit

The width of the internal data path of the CPU.

802.3

An Institute of Electrical and Electronics Engineers (IEEE) standard describing the physical and data link layers of a local area network based on bus topology and carrier sense multiple access/collision detect (CSMA/CD).

ac distribution

The method of distributing ac power in a cabinet.

application program

A program, such as a financial spreadsheet program, that performs an end-user task.

ANSI

American National Standards Institute.

ASCII

American Standard Code for Information Interchange. A set of 7- or 8-bit binary numbers representing the alphabet, punctuation, numerals, and other special symbols used in text representation and communications protocol.

barrel connector

A female connector for connecting two sections of ThinWire cable.

baud rate

The speed at which signals are serially transmitted along a communications line. One baud equals one bit/second.

binary

A number system that uses two digits—0 and 1. They are represented in system circuitry by two voltage levels. Programs are executed in binary form.

bit

A binary digit; the smallest unit of information in a binary system of notation, designated as a 0 or a 1.

BMA

The name assigned to Digital's suite of special-purpose, high-performance protocols; including LAT, LASTport/Disk, LASTport/Tape, and LASTport/PDQ.

boot

See bootstrap.

boot device

The device that is booted at system startup. See bootstrap and device.

bootable medium

A fixed disk or magnetic tape cartridge containing software (such as an operating system) that a bootstrap program can load into the system memory and execute.

bootstrap (or boot)

To attempt to load the operating system software from a mass storage device. An internal program makes the attempt.

building block

See SBB.

bus

A channel (a set of wires) along which communication signals in a computer system travel.

byte

A group of eight binary digits (bits). A byte is one-quarter of a VAX system longword.

caddy

A compact disc holder used to insert a compact disc into a compact-disc drive.

central processing unit (CPU)

The part of the system that controls the interpretation and execution of instructions.

client system

A computer system, connected to the same local area network as a server system, that connects to and uses services provided by the server.

coaxial cable

A 2-conductor, concentric, constant-impedance transmission cable.

communications line

A cable along which electrical signals are transmitted. Devices or systems that are connected by a communications line can share information and resources.

computer system

A combination of system hardware, software, and external devices that perform operations and tasks.

console mode

The state in which the computer is controlled from the console terminal. Console mode is indicated by the console prompt >>> on the monitor screen. (Compare to program mode.)

console prompt

A prompt used for communication between the system manager and the computer.

controller

A system component, usually a printed circuit board, that regulates the operation of one or more peripheral devices.

CPU

Abbreviation for central processing unit. See central processing unit.

daisy chain

A group of systems linked together sequentially.

data

A formal representation of information suitable for communication, interpretation, and processing by humans or computers.

data transmission

The movement of data in the form of electrical signals along a communications line.

debug

To detect, locate, and correct errors (bugs) in hardware or software.

DECnet

Digital's networking software that runs on nodes in both local and wide-area networks.

default

A value or setting used by the system if a specific setting is not entered by a user. Default settings are often set at the factory and reflect the most commonly used values.

DELNI

Digital's local network interconnect product that provides eight separate network interfaces from a single transceiver trap.

DEMPR

Digital's multiport repeater that provides eight ThinWire Ethernet drops from a single standard Ethernet connection.

deskside expansion unit

The StorageWorks floor-mounted stand that encloses a BA350-SA SBB.

device

The general name for any unit that is connected to the system and is capable of receiving, storing, or transmitting data. See *input/output* (I/O) device and controller.

device icon

An *icon* on the rear of the InfoServer 1000 that represents the device that can be plugged into the connector.

device name

The name by which a device or controller is identified in the system.

diagnostics

Programs, located in read-only memory, that detect and identify abnormal system hardware operation. See *read-only memory*.

disk

A flat circular plate with a coating that stores data magnetically in concentric circles (tracks). A disk resides permanently inside a disk drive, while a diskette is removable.

disk drive

A device that holds a *disk*. The drive contains mechanical components to spin the disk and move the read/write heads that store and read information on the surface of the disk.

disk server

A hardware system designed to provide operating system and data storage for other users.

DWZZA

The StorageWorks compatible SCSI bus signal converter.

error message

A message displayed by a system to indicate a mistake or malfunction.

Ethernet

A type of local area network based on carrier sense multiple access with collision detection (CSMA/CD).

faceplate

A wall receptacle that provides a network connection for a system.

file

A collection of related information treated by the system as a single unit.

format

To prepare a diskette to accept data.

formatted data

Data that is structured in a particular pattern to be understood by the system software.

full-height

In the storage industry, a device of conventional dimensions, a single device mounted in a 5.25-inch SBB.

ground

A voltage reference point in a system that has a zero voltage potential.

half-height

In the storage industry, a device of conventional dimensions, except for height. A device that occupies only one-half of a 5.25-inch storage bay.

hard disk

A disk that resides permanently inside a disk drive.

hardware

The physical components—mechanical and electrical—that make up a system. Compare to *software*.

head

The part of a fixed disk drive, diskette drive, or tape drive that reads, records, and erases data. Also called read/write head.

icon

A graphic symbol that visually represents a device or a procedure's function.

IEEE

Institute of Electrical and Electronics Engineers.

input/output (I/O) device

A piece of equipment that accepts data for transmission to (input) and from (output) the system. For example, a terminal.

I/O

Input/output. See input/output (I/O) device.

kernel

The integral part of the operating system that controls processes, system scheduling, memory management, input and output services, device management, network communications, and the organization of the file systems.

LAT

Local area transport. The LAT is a communications protocol that implements terminal emulation over an Ethernet local-area network. Terminals are normally connected to terminal servers, which send messages containing terminal data to host systems. The host systems view the remote terminal as if it was locally connected to the host.

LED

Light-emitting diode. LEDs are used as indicators on the system enclosure.

link

A communications path between two nodes. A physical link is the electrical connection between two nodes. A logical link implies that two nodes are able to communicate whether or not they have a direct physical link.

load

To copy software (usually from a peripheral device) to memory. To physically place a disk in a disk drive or a tape in a tape drive.

load device

The drive that holds the distribution media during software installation.

local area network (LAN)

A high-speed communications network that covers a limited geographical area; such as a section of a building, an entire building, or a cluster of buildings. It is a privately owned communication network whose speed is upward of 1 megabit/second.

logical link

Two nodes are able to communicate by means of software whether or not they have a direct physical link.

megabyte

1,048,576 (2²⁰) bytes.

memory

The area of the system that electrically stores instructions and data (often temporarily).

memory module

A printed circuit board that contains additional memory for the system.

module

A printed circuit board that contains electrical components and electrically conductive pathways between components. A module stores data or memory, or controls the functions of a device.

multiport repeater

A repeater used to connect two or more cable segments. The repeater extends Ethernet networks beyond the limits imposed by a single segment. Repeaters perform the basic actions of restoring signal amplitude, waveform, and timing amplitude to normal data and collision signals.

multisegment LAN

A LAN composed of more than one cable segment.

network

A group of individual computer systems that are connected by communications lines to share information and resources.

network coordinator

The network coordinator manages the network, assigns unique node names and addresses for each system on the network, and provides administrative assistance to network users.

network manager

See network coordinator.

node

An individual information-processing unit, such as a computer, workstation, or peripheral device, that is connected to a network. A node is an end point to any branch of a network or a junction common to two or more branches.

OPEN DECconnect

Digital's simple, cost-effective cabling system for extending Ethernet and terminal interconnections into offices and work areas.

OPEN DECconnect faceplate

See faceplate.

operating system

A collection of system programs that controls the operation of the system and allows the user access to data files, input/output devices, and applications programs. The operating system software performs such tasks as assigning memory to programs and data, processing requests, and scheduling jobs. VMS, ULTRIX, and MS-DOS are examples of operating systems.

output device

A device that accepts data from the system. A printer is an example of an output device.

peripheral device

A device that provides the CPU with additional memory storage or communications capability. Examples are disk and diskette drives, video terminals, and printers.

port

Another term for a connector on a device (such as a computer or server), used to make a connection to another device (such as a terminal, printer, or communications device). Connectors are usually on the rear of equipment.

power-on sequence

A series of ordered events that occur when you supply power by turning on the system.

program

The sequence of instructions the system uses to perform a task. See software.

program mode

The state in which the computer is controlled by the operating system. After the operating system is installed, the system always operates in program mode, unless put into console mode. See *console mode*.

prompt

Words or characters that the system displays to indicate that it is waiting for a command to be entered.

RAM

Abbreviation for random-access memory. See random-access memory.

random-access memory (RAM)

Memory that can be both read and written into, and can randomly access any one location during normal operations. The type of memory the system uses to store the instructions of programs currently being run.

read-only memory (ROM)

A memory whose contents cannot be modified. The system can use the data contained in a ROM, but cannot change the data.

ROM

Read-only memory.

run

n. A single continuous execution of a program. -v. To execute a program.

SBB

StorageWorks building block. The basic building block of the StorageWorks product line. Any device conforming to shelf mechanical and electrical standards installed in either a 3.50-inch or 5.25-inch carrier is considered to be an SBB, whether it is a storage device, a power supply, or other device.

SBB shelf

The common name for any StorageWorks shelf that contains only the power supply and storage SBBs.

SCSI

See Small Computer System Interface.

section

A single length of ThinWire Ethernet cable terminated at each end with a connector.

segment

A length of cable made up of one or more cable sections connected with barrel connectors or T-connectors.

server

Hardware or software that provides a specific set of services to a satellite.

single-segment LAN

A LAN that is composed of only one coaxial cable segment. See also section and segment.

Small Computer System Interface (SCSI)

An interface designed for connecting disks and other peripheral devices to computer systems. SCSI is defined by an American National Standards Institute (ANSI) standard and is used by many computer and peripheral vendors throughout the industry.

software

Programs executed by the system to perform a chosen or required function. Compare to *hardware*.

standard Ethernet

IEEE standard 802.3 compliant Ethernet network composed of standard Ethernet cable. Compare to *ThinWire Ethernet* cable.

storage medium

Any device capable of recording information, such as a hard disk.

StorageWorks

The Digital set of enclosure products that allows customers to design and configure their own storage subsystem. Components include power, packaging, and interconnections in a StorageWorks shelf. SBBs and array controllers are integrated to form storage subsystems. System-level enclosures to house the shelves and standard mounting devices for SBBs are also included.

store

To enter data into a storage device, such as a disk, or into memory.

system

A combination of system hardware, software, and peripheral devices that performs specific processing operations.

system management tasks

Tasks performed by an assigned person (usually the system manager) to operate and maintain the system.

T-connector

A three-point connector used to join ThinWire Ethernet cable sections. The connector also has a connector that is attached directly to a system.

terminal

An input/output device that lets a user communicate with the system. Terminals are divided into two categories—video and hardcopy.

terminator

A connector used on one or both ends of a communication line or bus that provides the 50-ohm termination resistance needed for the cable. Used on an Ethernet segment or a SCSI bus.

ThinWire

A Digital trademark used to describe its 10 base 2 (IEEE standard 802.3 compliant) Ethernet products used for local distribution of data.

ThinWire connector

The connector on the rear of the InfoServer 1000 to which the ThinWire Ethernet cable is attached.

ThinWire Ethernet

An *Ethernet* network that uses *ThinWire* cable. ThinWire is ideal for small standalone networks in which the users share resources, such as printers and storage devices.

transceiver

A device that provides a single physical connection between standard Ethernet and Ethernet communication equipment.

twisted-pair cable

Multiple-conductor cable whose component cables are paired, twisted, and enclosed in a single jacket.

user input device

A piece of equipment that is used to transfer data to the system. For example, keyboard, disk, tape, and system are input devices.

Index

A

AC power connector, 2-5, 2-12, 3-7 AC power selector switch setting, 3-15 ANSI standard SCSI, C-1

В

Barrel connector, 7-3
Baud rate communication port, 1-8 terminal, 2-10, 3-13, 4-12
BOOT command, 8-11
Booting, 5-1 to 5-4
Booting the server, 5-3 automatically with a console, 5-3 automatically without a console, 5-3
Booting the system with a console, 5-3

С

Cables SCSI, 6-2, 6-3 standard Ethernet, 7-10 ThinWire Ethernet, 7-3 transceiver, 7-10 troubleshooting, 8-2 Caddy, B-1 to B-6 ejecting manually, B-6 CD-ROM compact-disc drive connecting two, 6-5 expansion peripheral, 6-7 CD-ROM compact-disc drive (cont'd) resetting SCSI ID, 6-8 using, 6-7verifying SCSI ID, 6-7 CD-ROM services, 1–9 Commands summary, 8-17 Communications port, 1-8 baud rate, 1-8 Compact disc loading, B-1 preventive maintenance, 6-9 testing, 8-11 unloading, B-6 Compact-disc caddy, B-1 Compact-disc drive, 1-9 activity light, B-2 connecting one, 6-3 loading a disc, B-1SCSI ID numbers, 6-8 test disc utility, 8-11 unloading a disc, B-6 Configuration display, 6-8, 8-9 device mnemonics, 8-8 example, 8-9 Configurations customizing, C-1 for expansion peripherals, 6-1 Connecting console terminal, 2-10, 3-13, 4-12 Ethernet connectors, 2-5, 3-9, 4-13 expansion peripherals, 6-2, 6-3 two, 6-5 power cord, 2-12, 3-15 standard Ethernet networks, 7-9 to 7-11 Connecting (cont'd) ThinWire Ethernet networks, 7-2 to 7-4 to network, 5-2 transceiver cable, 7-10 Connectors, 2-5, 3-6, 4-4 AC power, 2-5, 3-7 console MMJ, 2-5, 3-7, 4-5 DC power, 2-5SCSI port, 2-5, 3-7 standard Ethernet, 2-5, 3-7 ThinWire Ethernet, 2-5, 3-7, 4-5 Console mode, 8-5, 8-11 Console prompt (>>>), 6-8 Console terminal, 1-10 Conventions used in this document, xiv Customizing SCSI ID numbers, C-1

D

DC power connector, 2-5 DECconnect cabling components, 7-5 **DECconnect** faceplate connecting, 7-5 DEMPR troubleshooting, 7-7 **Device** information displaying, 8-10 SHOW DEVICE command, 8-10 Device testing, 8-7 DEVNAM mnemonics. 8-8 Diagnostic indicators, 8-17 Diagnostics configuration display, 8-9 device, 8-7 Ethernet, 2-7, 2-9, 3-10, 3-12, 4-14 power-on display, 8-5 self-tests, 8-5, 8-7 standard Ethernet, 7-9 ThinWire Ethernet, 7-2 Diagnostic terminal, 1–10 Digital Services, 8–17

Dimensions InfoServer 1000, D-2 Disc drives turning on, 5-1

Ε

Environmental requirements, 2-1, 3-1, 4-1 Error messages configuration, 8-9 Ethernet cable, 8-7 power-on display, 8-5, 8-6 system, 8-9 Ethernet connectors, 2-5, 3-7 attaching to InfoServer 1000, 2-5 attaching to InfoServer SW1000, 4-13 attaching to InfoTower, 3-9 Ethernet diagnostics, 2-7, 2-9, 3-10, 3-12, 4-14 Expansion peripherals, 6-1 See also compact disc drive cables, 6-2 connecting one, 6-3 connecting two, 6-5 guidelines, 6-2 possible configurations, 6-1 power cord, 6-2, 6-3 See also the specific device, 6-1 signal cables, 6-2 supported number of, 6-1 terminator, 6-2 testing RRD42, 8-11 troubleshooting, 8-4 turning on, 2-14, 4-15, 5-1 unpacking, 6-2

F

Faceplate, OPEN DECconnect connecting, 7-5
Features InfoServer 1000 system, 1-8
Field-replaceable units, E-1 to E-3 InfoServer 1000, E-2 InfoTower, E-3 Field-replaceable units (cont'd) power cables, E-3

G

Grounding power strip, 2-2, 3-2, 4-2

Η

H4005 transceiver, 7-10 Handling problems with power-on display, 5-2 Hardware specifications InfoServer 1000, D-1 Humidity requirements, 2-1, 3-1, 4-1

Icons indicators, 1-10, 1-11 port and connector, 2-5, 3-6, 4-4 Indicators meaning, 1-10 troubleshooting, 8-17 InfoServer 1000 attaching Ethernet connectors, 2-5 choosing a location, 2-1connecting transceiver cable, 7-10 diagnostic indicators, 8–17 dimensions, D-2 Field-replaceable units, E-2hardware specifications, D-1 installation checklist, A-1 ports and connectors, 2-5 power requirements, D-2 storage conditions, D-3, D-11 troubleshooting, 8-3 unpacking, 2-3 InfoServer 1000 system features, 1-8 to 1-9 hardware components, 2-4overview, 1-1 to 1-9 possible applications, 1-1InfoServer Librarian installation, 3–1

InfoServer Librarian system, 3-2 InfoServer Scribe installation, 2-1 InfoServer SW1000 attaching Ethernet connectors, 4-13 choosing a location, 4-1 enclosure, 1-9 installation checklist, A-3 ports and connectors, 4-4 unpacking, 4-4 InfoServer SW1000 system features, 1-8 to 1-9 hardware components, 4-4 overview, 1-1 to 1-9 InfoServer VXT installation, 2-1 InfoTower attaching Ethernet connectors, 3–9 choosing a location. 3-1 connecting transceiver cable, 7–10 diagnostic indicators, 8-17 enclosure, 1-8 Field-replaceable units, E-3 installation checklist, A-2 ports and connectors, 3-6 troubleshooting, 8-3 turning on, 3-17 unpacking, 3-3 InfoTower system features, 1-8 to 1-9 hardware components, 3-5 overview, 1-1 to 1-9 possible applications, 1-1InfoTower system installation, 3-1 to 3-17 Inserting compact disc, B-1 Installation console terminal, 2-10, 3-13, 4-12 Ethernet T-connector and terminators, 2-5, 3-9, 4-13 expansion peripherals, 6-1 InfoServer 1000 power supply, 2-12 location, 2-1, 3-1, 4-1 loopback plug, 2-5, 3-9 power cord, 2-12, 3-15 SCSI terminator, 3-8 unpacking InfoTower system, 3-2 unpacking standalone system, 2-3

Installation (cont'd) unpacking StorageWorks system, 4-3 voltage selection, 3-15 Installation checklist InfoServer 1000, A-1 InfoServer SW1000, A-3 InfoTower, A-2

K

Kernel, 1–11

L

Loading compact disc, B-1 Loopback plug attaching to InfoServer 1000, 2-5 attaching to InfoTower, 3-9 diagnostics, 2-9, 3-12 in network diagnostics, 7-9

Μ

Memory, 1-8 MMJ console port, 2-5, 3-7, 4-5 Moving the system, 2-3, 3-6, 4-4

N

Networks, 7-1 to 7-11 connecting, 5-2

0

On/off switch on (|) position, 3-17 OPEN DECconnect cabling components, 7-5 connecting faceplate, 7-5 Operating conditions system specifications, D-3, D-11 Operating system software restarting, 8-11 |/O power switch on (|) position, 5-1 Outlets, grounded, 2-2, 3-2, 4-2

Ρ

Password, 5-2, 5-4, 8-4, 8-13 Peripheral devices, 6-1 installing. 6-1 See also the specific device, 6-1supported number of, 6-1 Ports, 2-5, 3-6, 4-4 baud rate, 1-8 Power connector, AC InfoServer 1000, 2-5 InfoTower, 3-6 Power connector, DC InfoServer 1000, 2-5 Power cord, AC connecting, 2-12, 3-15 expansion peripheral, 6-2, 6-3, 6-5 Power cord, DC connecting, 2-12 Powering on and off InfoTower, 3–17 standalone, 2-14 StorageWorks, 4–15 Power-on display, 5-1 description of identifiers in, 8-6 error messages, 8-5, 8-6 examples of common status messages in, 8-6 troubleshooting, 5-2, 8-3 Power requirements InfoServer 1000, D-2 Power selector switch setting. 3-15 Power strip, 2-2, 3-2, 4-2 Power supply, 1-8 Program mode, 8-11

Q

Question marks in power-on display, 8–6

R

Removing compact disc, B-6 Restarting system, 8-11 RRD42 compact-disc drive troubleshooting, 8-4 Rules for setting SCSI ID numbers, C-4

S

SCSI backplane, 4-5 SCSI bus, 1-8, C-1 device, C-1 expansion peripheral cables, 6-2 terminating, 4-5 terminator, 3-8, 6-2, 6-3, C-4 SCSI ID numbers CD-ROM, 6-8 creating custom configurations, C-1 default settings, C-1 definition of, C-1 rules for setting, C-4 setting, C-1 using SHOW DEVICE command to display, 6-8 SCSI port, 2-5, 3-7 connecting expansion peripheral, 6-2 Segment ThinWire Ethernet, 7–3 Self-tests, 8-7 restarting after, 8-11 Server system software troubleshooting, 8-3 Service information, 8-17 SET commands SET XDELTA, 8-5 summary of, 8-17 Settings SCSI ID numbers, 6-8 terminal, 2-10, 3-13, 4-12 Shipping cartons unpacking InfoTower system, 3-2

Shipping cartons (cont'd) unpacking standalone InfoServer 1000 system, 2–3 unpacking StorageWorks InfoServer SW1000 system, 4-3 SHOW commands SHOW CONFIG. 8-7.8-9 SHOW DEVICE, 6-8, 8-10 SHOW ERROR, 8-8 summary of, 8-17 Signal cables expansion peripheral, 6-2Site preparation, 2-1, 3-1, 4-1Software troubleshooting, 8-4 Standalone InfoServer 1000 turning on, 2-14 Standalone system installation, 2-1 to 2-14 Standard Ethernet attaching loopback plug, 2-5, 3-9 transceiver cable, 7–10 troubleshooting, 7-11 Standard Ethernet connector, 2-5, 3-7 Standard Ethernet network, 1-8 connecting to, 7-9 to 7-11 verifying, 7-9 Starting the system, 5-1 Storage conditions InfoServer 1000, D-3, D-11 StorageWorks InfoServer SW1000 turning on, 4-15 StorageWorks pedestal, 4-10, 4-11 StorageWorks system installation, 4-1 to 4-16 Summarv of SET commands, 8–17 of SHOW commands, 8-17 Surge protector, 2–2, 3–2, 4–2 Switches AC power for InfoTower enclosure, 3–17 AC power selector, 3-15 System configuration display, 8-9 connecting console terminal, 2-10, 3-13, 4 - 12

System (cont'd) connecting power cord, 2-12, 3-15 connecting to network, 5-2 System configuration displaying with SHOW DEVICE command, 6-8 System SCSI ID numbers, C-2 System specifications operating conditions, D-3, D-11

T

T commands, 8-7 T-connector attaching terminators, 2-7, 3-10, 4-14 attaching to InfoServer 1000, 2-5, 7-3 attaching to InfoServer SW1000, 4-13, 7-3 attaching to InfoTower, 3-9, 7-3 connecting ThinWire Ethernet cable, 7-3 diagnostics, 2-7, 3-10, 4-14 removing from InfoServer 1000, 7–7 troubleshooting, 7-7 Temperature requirements, 2-1, 3-1, 4-1, D-3, D-11 Terminal baud rate, 2-10, 3-13, 4-12 connecting, 2-10, 3-13, 4-12 optional console, 1-10 settings, 2–10, 3–13, 4–12 troubleshooting, 8-3 turning on, 5-1 Terminators attaching to T-connector, 2-7, 3-10, 4-14 SCSI, 6-2, C-4 ThinWire Ethernet, 7-3 troubleshooting, 7-7 TEST commands, 8–7 T NI, 7-2, 7-9, 8-7, 8-8 T SCSI, 8-11 Test disc utility, 8–11 ThinWire Ethernet attaching T-connector, 2-5, 3-9, 4-13

ThinWire Ethernet cable connecting, 7-3 to create a workgroup, 7-5 ThinWire Ethernet connector, 2-5, 3-7, 4-5 ThinWire Ethernet network, 1-8 connecting to, 7-2 to 7-4removing systems from, 7-7 troubleshooting, 7-7 verifying, 7–2 ThinWire Ethernet segment, 7–3 T NI command, 7-2, 7-9 Transceiver cable, 7–10 Troubleshooting, 8-1 to 8-18 configuration display, 8–9 expansion peripheral, 8-4 indicators (LEDs), 8-17 InfoServer 1000, 8-3 power-on display, 5-2RRD42, 8–4 self-tests, 8-7 software, 8-4 standard Ethernet, 7–11 terminal, 8-3 ThinWire Ethernet, 7-7 troubleshooting table, 8-3 Turning power on, 5–1 Turning power on and off InfoTower, 3-17 standalone, 2–14 StorageWorks, 4-15 Turning the system on and off InfoTower, 3-17 standalone, 2-14 StorageWorks, 4-15

U

Unloading compact disc, B-6 Unpacking expansion peripheral, 6-2 InfoTower system, 3-2 standalone InfoServer 1000, 2-3 StorageWorks InfoServer SW1000, 4-3

V

Ventilation, 3-2 Voltage selection InfoTower, 3-15

W

Workgroup creating, 7–5

How to Order Additional Documentation

Technical Support

If you need help deciding which documentation best meets your needs, call 800-DIGITAL (800-344-4825) and press 2 for technical assistance.

Electronic Orders

If you wish to place an order through your account at the Electronic Store, dial 800-234-1998, using a modem set to 2400- or 9600-baud. You must be using a VT terminal or terminal emulator set at 8 bits, no parity. If you need assistance using the Electronic Store, call 800-DIGITAL (800-344-4825) and ask for an Electronic Store specialist.

Telephone and Direct Mail Orders

From	Call	Write
U.S.A.	DECdirect Phone: 800-DIGITAL (800-344-4825) Fax: (603) 884-5597	Digital Equipment Corporation P.O. Box CS2008 Nashua, NH 03061
Puerto Rico	Phone: (809) 781-0505 Fax: (809) 749-8377	Digital Equipment Caribbean, Inc. 3 Digital Plaza, 1st Street Suite 200 Metro Office Park San Juan, Puerto Rico 00920
Canada	Phone: 800-267-6215 Fax: (613) 592-1946	Digital Equipment of Canada Ltd. 100 Herzberg Road Kanata, Ontario, Canada K2K 2A6 Attn: DECdirect Sales
International		Local Digital subsidiary or approved distributor
Internal Orders ¹ (for software documentation)	DTN: 264-3030 (603) 884-3030 Fax: (603) 884-3960	U.S. Software Supply Business Digital Equipment Corporation 10 Cotton Road Nashua, NH 03063-1260
Internal Orders (for hardware documentation)	DTN: 264-3030 (603) 884-3030 Fax: (603) 884-3960	U.S. Software Supply Business Digital Equipment Corporation 10 Cotton Road Nashua, NH 03063-1260

¹Call to request an Internal Software Order Form (EN-01740-07).

.

Reader's Comments

InfoServer 1000 Installation and Owner's Guide EK-INFLC-OM. B01

Your comments and suggestions help us improve the quality of our publications. Thank you for your assistance.

I rate this manual's:	Excellent	Good	Fair	Poor
Accuracy (product works as manual says) Completeness (enough information)				
Clarity (easy to understand)				
Viganization (structure of subject matter)				
Figures (useful) Examples (useful)	п			
Index (ability to find topic)		П		Π
Page layout (easy to find information)				
I would like to see more/less				
What I like best about this manual is				
What I like least about this manual is				
I found the following errors in this manual Page Description	l:		•	
Additional comments or suggestions to imp	prove this ma	nual:		
		· · · · · · · · · · · · · · · · · · ·		
For software manuals, please indicate whi	ch version of	the softwar	e you are us	ing:
			<u></u>	
Name/Title		Dept.	<u> </u>	
Company			Date	
Mailing Address				
		_ Phone		

Do Not Tear - Fold Here and Tape





FIRST CLASS PERMIT NO. 33 MAYNARD MASS.

POSTAGE WILL BE PAID BY ADDRESSEE

DIGITAL EQUIPMENT CORPORATION Information Design and Consulting ZKO1–3/J35 110 SPIT BROOK RD NASHUA, NH 03062–9987

[BeenHellinedIneedellichelistedellist

No Postage

Necessary If Mailed in the United States

Do Not Tear - Fold Here -

Reader's Comments

InfoServer 1000 Installation and Owner's Guide EK-INFLC-OM. B01

Your comments and suggestions help us improve the quality of our publications. Thank you for your assistance.

I rate this manual's:	Excellent	Good	Fair	Poor	
Accuracy (product works as manual says) Completeness (enough information) Clarity (easy to understand) Organization (structure of subject matter) Figures (useful) Examples (useful) Index (ability to find topic) Page layout (easy to find information) I would like to see more/less $A fin MMJ Convertor V set for What I like best about this manual is$	int figu Ne +ty	- - - - - - - - - - - - - - - - - - -	1/e 2 2000	DECIS al cons	sole)
what I like best about this manual is					
What I like least about this manual is					
I found the following errors in this manual Page Description <u>f-8</u> <u>bullet 5 Section 2-3.4 i</u>	1: n- 045 6u	de docon	texit-		
Additional comments or suggestions to imp	prove this ma	nual:			
For software manuals, please indicate whi	ch version of	the softwar	e you are us	sing:	
Name/Title		Dept.			
Company			Date		
Mailing Address					
		_ Phone			

Do Not Tear - Fold Here and Tape



BUSINESS REPLY MAIL

FIRST CLASS PERMIT NO. 33 MAYNARD MASS.

POSTAGE WILL BE PAID BY ADDRESSEE

DIGITAL EQUIPMENT CORPORATION Information Design and Consulting ZKO1-3/J35 110 SPIT BROOK RD NASHUA, NH 03062-9987

Hhmallallandlandallalahahahahahahahah

No Postage

Necessary If Mailed in the United States

1

Do Not Tear - Fold Here -

digital